

How aggregate volatility-of-volatility affects stock returns

Article

Accepted Version

Hollstein, F. and Prokopczuk, M. (2018) How aggregate volatility-of-volatility affects stock returns. The Review of Asset Pricing Studies, 8 (2). pp. 253-292. ISSN 2045-9939 doi: https://doi.org/10.1093/rapstu/rax019 Available at http://centaur.reading.ac.uk/75564/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1093/rapstu/rax019

Publisher: Society for Financial Studies

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other copyright holders. Terms and conditions for use of this material are defined in the <u>End User Agreement</u>.

www.reading.ac.uk/centaur

CentAUR

Central Archive at the University of Reading



Reading's research outputs online

How Aggregate Volatility-of-Volatility Affects Stock Returns*

Fabian Hollstein[†] and Marcel Prokopczuk^{†,‡}

January 18, 2017

Abstract

A stylized theoretical model with stochastic volatility suggests the existence of a trade-off between returns and volatility-of-volatility. Using the VVIX index as empirical measure, we confirm this prediction and detect that time-varying aggregate volatility-of-volatility commands an economically substantial and statistically significant negative risk premium. We find a two-standard deviation increase in aggregate volatility-of-volatility factor loadings to be associated with a decrease in average annual returns of about 11%. This phenomenon cannot be explained by aggregate volatility, jump risk, and several other characteristics and factor sensitivities, or by a crisis effect.

JEL classification: G12, G11

Keywords: Aggregate economic uncertainty, stock market volatility-of-volatility, VVIX

^{*}We are grateful to Zeno Adams, Jan Antell (discussant), I-Hsuan Ethan Chiang (discussant), David Feldman (discussant), Roland Füss, Anastasios Kagkadis, Paola Pederzoli (discussant), Michael Weber, and Chardin Wese Simen, as well as seminar participants at the meetings of the Eastern Finance Association, FMA Europe, Midwest Finance Association, Swiss Society for Financial Market Research and several seminars for valuable comments. We thank Chen Xue for kindly sharing his data. Contact: hollstein@fmt.unihannover.de (F. Hollstein) and prokopczuk@fmt.unihannover.de (M. Prokopczuk).

[†]School of Economics and Management, Leibniz University Hannover, Koenigsworther Platz 1, 30167 Hannover, Germany.

[‡]ICMA Centre, Henley Business School, University of Reading, Reading, RG6 6BA, UK.

I Introduction

In this paper, we study whether aggregate volatility-of-volatility influences stock returns.¹ Aggregate volatility-of-volatility is important because it characterizes the distribution of stochastic volatility. While there seems to be a consensus that stochastic volatility is important for asset pricing (e.g., Ang et al., 2006b; Barndorff-Nielsen & Veraart, 2012), volatility-of-volatility, thus far, has received considerably less attention. However, the volatility of stochastic volatility likely also carries important information on the current and future investment opportunity set. Hence, if an innovation to aggregate volatility-of-volatility predicts a worsening of future investment opportunities, investors seek to insure against this, demanding stocks with low sensitivities to innovations in volatility-of-volatility.

Using a simple stylized theoretical model with recursive preferences and consumption uncertainty in the spirit of Bollerslev et al. (2009) we motivate the existence of a volatilityof-volatility-return trade-off in addition to the commonly employed risk-return and volatilityreturn trade-offs. Based on the model predictions, we show empirically that volatility-ofvolatility is indeed significantly priced in the stock market.

We document the trade-off between volatility-of-volatility and returns using several steps. First, we show that market volatility-of-volatility, measured using the VVIX index, is distinct from market volatility. The shocks to the two measures are only moderately correlated (0.67) and the overall dynamics appear to be substantially distinct. Moreover, empirically, the VIX forecasts future market volatility, while the VVIX forecasts the future volatility of the VIX. Second, following the empirical framework of Ang et al. (2006a) and Cremers et al. (2015), we show that the single-sorted hedge portfolio on innovations in aggregate volatility-of-volatility yields an annual return of -4.91% and an alpha of -6.35% with respect to the factor model of Hou et al. (2015b).² Additionally, using Fama & MacBeth (1973) cross-sectional regressions, we find that a two-standard deviation increase in aggregate volatility-of-volatility factor loadings is associated with a significant decrease in average annual returns of about 11%. We show that these results hold both for value-weighted and equally weighted sorts and

¹Aggregate volatility-of-volatility denotes the volatility of the volatility of the aggregate stock market.

²The alpha with respect to the classical Carhart (1997) 4-factor model is -4.91%, which is also significantly different from zero.

regressions and prevail after controlling for canonical characteristics and popular market risk factors, including, e.g., beta, aggregate volatility, size, liquidity, and jump risk.

Third, we show that our findings are robust to various additional tests. Performing multivariate sensitivity estimations to control for various aggregate factors, our conclusions are unchanged. Likewise, we find that our results are not driven by the 2008–2009 financial crisis. Lastly, we show that innovations in aggregate volatility-of-volatility can be hedged using options written on the volatility index.

Our work is related to recent studies on uncertain volatility. Epstein & Ji (2013) formulate a multiple priors model with ambiguity about volatility. Our approach is similar in that we consider uncertainty about volatility. However, we use a direct smooth measure to capture this uncertainty. Huang & Shaliastovich (2014) show that there is a volatilityof-volatility risk premium in the cross-section of delta-hedged S&P 500 and VIX options. Unlike Huang & Shaliastovich (2014) we study the pricing of aggregate volatility-of-volatility in the cross-section of equity returns. Studying the cross-section of stock returns allows us to control for several other cross-sectional risk factors, which is not easily possible when working with hedged options portfolios. Baltussen et al. (2015) show that the volatility-ofvolatility of individual stocks is significantly negatively priced in the cross-section of stock returns. As opposed to our study, Baltussen et al. (2015) primarily use a measure of idiosyncratic volatility-of-volatility and cannot detect a significant effect using past sensitivities from aggregate factor specifications with high-minus-low idiosyncratic volatility-of-volatility portfolios or the volatility-of-volatility from at-the-money (ATM) S&P 500 options. We, in turn, examine model-free aggregate market volatility-of-volatility, represented by the VVIX, as a state variable concentrating on systematic instead of idiosyncratic effects. Chen et al. (2014) measure volatility-of-volatility using high-frequency index option data and empirically find volatility-of-volatility to carry a significantly negative risk premium. Our study differs from theirs in two important dimensions. First, we directly use the VVIX index provided by the CBOE, instead of a high frequency intraday realized variance measure of the VIX index. Consequently, we use a forward-looking volatility measure instead of past variation in forward-looking volatility. This is important since there appears to be a consensus in the literature, that using implied instead of historical volatility estimates significantly improves prediction accuracy (e.g., Jiang & Tian, 2005; Prokopczuk & Wese Simen, 2014). Second, we present a model framework with recursive Epstein–Zin preferences and consumption uncertainty that rationalizes our empirical results.

Our work also connects to the literature on aggregate risk factors including Ang et al. (2006b), Adrian & Rosenberg (2008), Han & Zhou (2012), Chang et al. (2013), Lettau et al. (2014), and Cremers et al. (2015). However, our focus is different. While previous studies test whether aggregate volatility, its components, or higher moments are priced, we complement the existing literature by examining the implications of aggregate volatility-of-volatility for the stock market.

Finally, our paper relates to the literature on long-run risks pioneered by Bansal & Yaron (2004). Our empirical model especially builds on Bollerslev et al. (2009), Campbell et al. (2014) and Bali & Zhou (2015). We show that with the model assumptions commonly employed in the literature, especially those of more recent works, the expected return beta representation implies that volatility-of-volatility is priced in asset markets.

The remainder of this paper is organized as follows. Section II presents a simple model that implies a trade-off of aggregate volatility-of-volatility and stock returns. Section III describes our dataset and methodology. Section IV presents our empirical results with portfolio sorts and cross-sectional regressions. Section V presents further analyses and robustness tests. Finally, Section VI concludes. Appendices A and B provide detailed descriptions of all variables used in the paper.

II Model Formulation

We build on the results of Bollerslev et al. (2009), Campbell et al. (2014), and Bali & Zhou (2015) and impose a stylized intertemporal asset pricing model with stochastic volatility to theoretically motivate the volatility-of-volatility-return trade-off.

The representative agent has Epstein & Zin (1989) preferences with the value function $V_{g,t}$ as

$$V_{g,t} = \left[(1-\delta) C_t^{\frac{1-\gamma}{\theta}} + \delta \left(E_t \left[V_{g,t+1}^{1-\gamma} \right] \right)^{\frac{1}{\theta}} \right]^{\frac{\theta}{1-\gamma}},$$
(1)

where C_t is the consumption at time t, and the preference factors of the representative agent are denoted by δ , the subjective discount factor, and γ , the coefficient of relative riskaversion. As is commonly done, for convenience we define $\theta = (1 - \gamma) / (1 - 1/\psi)$, with ψ being the elasticity of intertemporal substitution. As shown by Epstein & Zin (1991), the corresponding stochastic discount factor (SDF) can be expressed as

$$M_{t+1} = \left(\delta\left(\frac{C_t}{C_{t+1}}\right)^{1/\psi}\right)^{\theta} \left(\frac{W_t - C_t}{W_{t+1}}\right)^{1-\theta}.$$
(2)

 W_t is the market value of the agent's consumption stream. The logarithm of the SDF is then

$$m_{t+1} = \theta \ln \delta - \frac{\theta}{\psi} g_{t+1} + (\theta - 1) r_{t+1}.$$

$$(3)$$

 $r_{t+1} = \ln (W_{t+1}/(W_t - C_t))$ is the log return on wealth and $g_{t+1} = \Delta c_{t+1}$ is the log consumption growth. We follow Bollerslev et al. (2009) and Bali & Zhou (2015) assuming the following joint dynamics for consumption growth and consumption growth volatility

$$g_{t+1} = \mu_g + \sqrt{\sigma_{g,t}^2} z_{t+1},$$
 (4)

$$\sigma_{g,t+1}^2 = a_\sigma + \rho_\sigma \sigma_{g,t}^2 + \sqrt{q_t} z_{v,t+1}, \qquad (5)$$

$$q_{t+1} = a_q + \rho_q q_t + \varphi_q \sqrt{q_t} z_{q,t+1}.$$
(6)

 μ_g is the constant mean growth rate, $\sigma_{g,t}^2$ denotes the conditional variance of consumption growth, q_t represents the volatility uncertainty process, while z_{t+1} , $z_{v,t+1}$, and $z_{q,t+1}$ describe independent i.i.d. N(0,1) processes. The parameters satisfy $a_{\sigma} > 0$, $a_q > 0$, $|\rho_{\sigma}| < 1$, $|\rho_q| < 1$, and $\varphi_q > 0$.

Let ω_t denote the logarithm of the price-dividend ratio, or price-consumption or wealthconsumption ratio, of the asset that pays the consumption endowment. To find the equilibrium, one can conjecture a solution for ω_t as an affine function of the state variables $\sigma_{g,t}^2$ and q_t ,

$$\omega_t = A_0 + A_\sigma \sigma_{g,t}^2 + A_q q_t. \tag{7}$$

Using the standard Campbell & Shiller (1988) approximation $r_{t+1} = \kappa_0 + \kappa_1 \omega_{t+1} - \omega_t + g_{t+1}$,

a solution for the coefficients A_0 , A_σ , and A_q can be obtained (Bollerslev et al., 2009). Note that, assuming $\gamma > 1$ and $\psi > 1$, hence $\theta < 0$, it holds that both $A_\sigma < 0$ and $A_q < 0$. Substituting the Campbell & Shiller (1988) approximation into Equation (3), one obtains a pricing kernel without reference to consumption growth (Campbell et al., 2014; Bali & Zhou, 2015)

$$m_{t+1} = \theta \ln \delta + \frac{\theta}{\psi} \kappa_0 - \frac{\theta}{\psi} \omega_t + \frac{\theta}{\psi} \kappa_1 \omega_{t+1} - \gamma r_{t+1}.$$
(8)

Assuming a conditional joint log-normal distribution with time-varying volatility for the asset returns, the risk premium on any asset j is given by

$$E_t(r_{j,t+1}) - r_{f,t} + \frac{1}{2} Var_t(r_{j,t+1}) = -Cov_t(m_{t+1}, r_{j,t+1}).$$
(9)

Inserting the pricing kernel without reference to consumption growth in Equation (8) into Equation (9), one can obtain an ICAPM pricing relation of the following form:

$$E_t(r_{j,t+1}) - r_{f,t} + \frac{1}{2} Var_t(r_{j,t+1}) = \gamma Cov_t(r_{t+1}, r_{j,t+1}) - \frac{\theta}{\psi} \kappa_1 Cov_t(\omega_{t+1}, r_{j,t+1}).$$
(10)

Inserting Equation (7) into Equation (10) yields:

$$E_{t}(r_{j,t+1}) - r_{f,t} + \frac{1}{2} Var_{t}(r_{j,t+1}) = \gamma Cov_{t} (r_{t+1}, r_{j,t+1}) - \frac{\theta}{\psi} \kappa_{1} A_{\sigma} Cov_{t} (\sigma_{g,t+1}^{2}, r_{j,t+1}) - \frac{\theta}{\psi} \kappa_{1} A_{q} Cov_{t} (q_{t+1}, r_{j,t+1}).$$
(11)

Apart from the variance term $Var_t(r_{j,t+1})$, there is the usual risk-return trade-off γ , a trade-off of returns and variance $-\frac{\theta}{\psi}\kappa_1 A_{\sigma}$, and a volatility-of-volatility-return trade-off $-\frac{\theta}{\psi}\kappa_1 A_q$ can be detected from this formulation. This formulation may already directly serve as a motivation for why both shocks to volatility and volatility-of-volatility should be priced in the stock market. However, in this formulation, $\sigma_{g,t+1}^2$ and q_{t+1} represent the moments of the consumption processes.

The model also makes direct predictions on the relation of the expected return on wealth and the consumption moment. From this, one can directly infer information on the variance of the return on aggregate wealth (Bollerslev et al., 2009) as well as on volatility-of-volatility on aggregate wealth. We use the equations $\sigma_{r,t}^2 \equiv Var_t(r_{t+1}) = \sigma_{g,t}^2 + \kappa_1^2 \left(A_{\sigma}^2 + A_q^2 \varphi_q^2\right)$ and $Q_t \equiv Var_t(\sigma_{r,t+1}^2) = q_t \left(1 + \varphi_q^2 \kappa_1^4 \left(A_{\sigma}^2 + A_q^2 \varphi_q^2\right)^2\right)$, to obtain the following relation:

$$E_{t}(r_{j,t+1}) - r_{f,t} + \frac{1}{2} Var_{t}(r_{j,t+1}) = \gamma Cov_{t} (r_{t+1}, r_{j,t+1}) - \frac{\theta}{\psi} \kappa_{1} A_{\sigma} Cov_{t} (\sigma_{r,t+1}^{2}, r_{j,t+1}) + \frac{\theta}{\psi} \frac{\kappa_{1} \left[A_{\sigma} \kappa_{1}^{2} \left(A_{\sigma}^{2} + A_{q}^{2} \varphi_{q}^{2} \right) - A_{q} \right]}{\left[1 + \varphi_{q}^{2} \kappa_{1}^{4} \left(A_{\sigma}^{2} + A_{q}^{2} \varphi_{q}^{2} \right) \right]} Cov_{t} (Q_{t+1}, r_{j,t+1}) .$$
(12)

Equation (12) directly predicts a relation of current excess returns with variance and volatility-of-volatility of aggregate wealth. Note, however, that technically, $Cov_t (\sigma_{r,t+1}^2, r_{j,t+1})$ and $Cov_t (Q_{t+1}, r_{j,t+1})$ denote the time-t covariances of asset returns with t + 1 expectations of variance and volatility-of-volatility over time t + 2. Hence, for these higher-order terms to be estimable and the model to be directly applicable, we have to assume that the current expectation of aggregate variance and aggregate volatility-of-volatility provide good proxies for next-period expectations.³ Further, note that Equation (12) implies a relation of the variance as well as, technically, the variance-of-variance of total wealth with stock returns. In our empirical tests, we use shocks to the empirically more robust measures of aggregate volatility and aggregate volatility-of-volatility (e.g., Merton, 1980). However, shocks to both volatility and squared volatility always have the same sign and are highly correlated by construction. Hence, for the purpose of motivating our empirical analysis, although not perfectly similar, the model delivers a reasonable description.

Transforming Equation (12) to a conditional beta representation, replacing the conditional covariances with conditional betas, we get

$$E_t(r_{j,t+1}) - r_{f,t} + \frac{1}{2} Var_t(r_{j,t+1}) = \tilde{Y} \cdot \beta_{j,t}^{M} + \tilde{V} \cdot \beta_{j,t}^{V} + \tilde{Z} \cdot \beta_{j,t}^{Q}.$$
 (13)

 \tilde{Y} , \tilde{V} , and \tilde{Z} denote the risk premia, while $\beta_{j,t}^{M}$, $\beta_{j,t}^{V}$, and $\beta_{j,t}^{Q}$ are the conditional market, variance, and volatility-of-volatility betas.

From Equation (12), one can infer the model predictions on the market prices of risk.

³In our empirical study, we use daily observations. Our proxies for aggregate volatility and volatilityof-volatility denote expectations over the next 30 days. Hence, due to the large overlap of our empirical measures, this assumption appears reasonable.

The market prices of risk are proportional to the derivative of the negative of the stochastic discount factor to shocks in the state variables. It can be easily shown that the market price of risk of shocks to aggregate wealth is positive, and, given that the parameters satisfy $\psi > 1$, $\theta < 0$, and $A_{\sigma} < 0$ as well as $A_q < 0$, the market price of shocks to variance is negative. The market price of risk on aggregate volatility-of-volatility is not clearly signed and depends on the relation of $A_{\sigma}\kappa_1^2 \left(A_{\sigma}^2 + A_q^2\varphi_q^2\right)$ and A_q .

III Data and Methodology

A Data

We base our study on all stocks traded on the New York Stock Exchange (NYSE), the American Stock Exchange (AMEX), and the National Association of Securities Dealers Automated Quotations (NASDAQ) that are classified as ordinary common shares (Center for Research in Security Prices (CRSP) share codes 10 or 11), excluding closed-end funds and REITS (SIC codes 6720–6730 and 6798), for the sample period between March 01, 2006 and September 30, 2016. We obtain data on the VIX and the VVIX from the CBOE. The VIX is constructed so that it represents the model-free 30-day implied volatility of the S&P 500 index. On February 24, 2006, the CBOE began trading options written on the VIX and recently the CBOE started reporting the VVIX, which represents the model-free 30-day implied volatility of the VIX.^{4,5} We also obtain five-minute intraday high-frequency data on the VIX and the S&P 500 from the Thompson Reuters Tick History (TRTH) database.

We obtain daily and monthly price data as well as data on dividend payments, trading volumes, firm age, and shares outstanding from CRSP. Following Amihud (2002) and Zhang (2006), we exclude "penny stocks" with prices below 1 dollar. Additionally, we require a market capitalization of at least 225 million dollars (D'Avolio, 2002; Baltussen et al., 2015). These two thresholds serve to eliminate the most illiquid stocks that exhibit potential

 $^{^4{\}rm For}$ the period prior to 2007, we compute the VVIX ourselves using the VIX methodology and options on the VIX from OptionMetrics.

⁵For reliable implied moments, option liquidity is an important issue. The average daily trading volume of VIX options was already greater than 8,000 in February 2006. It increased gradually to more than 500,000 contracts per day towards the end of the sample. For more information, refer to the CBOE homepage.

microstructure problems and may bias the results (Fama & French, 2008). Furthermore they ensure that only stocks with relatively low short-sale constraints are selected (D'Avolio, 2002). On average, these selection criteria result in about 2,396 stocks included in our sample per month.⁶ We adjust for delisting returns following Shumway (1997) and Shumway & Warther (1999).

We collect balance sheet and income statement data from the Compustat database. Options data are from the IvyDB OptionMetrics database.⁷ Data on the Fama & French (1993) and momentum factors as well as the risk-free (Treasury Bill) rate are from Kenneth French's data library. We obtain data on the Pastor & Stambaugh (2003) liquidity factor from Robert Stambaugh's homepage and data on the Hou et al. (2015b) factors directly from Chen Xue.⁸

We provide detailed variable definitions of all variables used in the paper in Appendices A and B.

B Empirical Framework

Our goal is to test whether stocks with different sensitivities to innovations in aggregate volatility-of-volatility have different returns. To examine that, we follow a large body in the asset pricing literature, and examine the contemporaneous relation between realized factor loadings and realized returns (e.g., Black et al., 1972; and Fama & French, 1992; among many others). Ang et al. (2006a) argue that while pre-formation factor loadings reflect both actual variation in factor loadings and measurement error, post-formation factor loadings are almost exclusively affected by stock return covariations with risk factors. Additionally, they point out that if risk exposures, and hence factor loadings, are highly time-varying, pre-formation factor loadings might be poor predictors of ex post risk exposures, leaving the analysis with low power to detect relations between factor loadings and realized returns. Addressing these concerns, our research design follows Ang et al. (2006a) and Cremers et al.

 $^{^{6}}$ The minimum is at 1,807 stocks eligible in February 2009, while the maximum results in April 2006, with 2,794 stocks. The total number of firm-month observations is 280,333.

⁷Options data has only been available up to April 30, 2016 when we revised this manuscript. Hence, all tests that include options data (e.g., jump, dSkew) are performed for the sample period until April 30, 2016.

⁸The data from Chen Xue is monthly and only ranges until December 2015. Therefore, we construct the factors ourselves following the description in Hou et al. (2015a) to add the data for January until September 2016 as well as to obtain daily returns on the factors.

(2015) by estimating factor loadings for individual stocks using daily returns over rolling annual periods from the regression:

$$r_{j,\tau} - r_{f,\tau} = \alpha_{j,t} + \beta_{j,t}^{\mathrm{M}}(r_{M,\tau} - r_{f,\tau}) + \beta_{j,t}^{\mathrm{V}}dVIX_{\tau} + \beta_{j,t}^{\mathrm{Q}}dVVIX_{\tau} + \epsilon_{j,\tau}.$$
 (14)

 $r_{j,\tau}$ is the daily return of asset j on day τ , $r_{M,\tau}$ is the return of the market on that day, and $r_{f,\tau}$ is the risk-free rate. $dVIX_{\tau}$ and $dVVIX_{\tau}$ are the daily innovation in the VIX (Ang et al., 2006b) and VVIX indices.

To mirror the specification implied by our theoretical model in Equation (13), the regression in Equation (14) only includes the market excess return, innovations in aggregate volatility, and innovations in aggregate volatility-of-volatility. Furthermore, Ang et al. (2006b) argue that including additional factors in the regression in Equation (14) may add a lot of noise. We control for further factors when performing the time series and cross-sectional asset pricing tests. As a robustness check and to account for possible model mis-specification, we also consider multivariate joint factor loading estimations controlling for several aggregate risk factors previously documented in the literature:

$$r_{j,\tau} - r_{f,\tau} = \alpha_{j,t} + \beta_{j,t}^{\mathrm{M}}(r_{M,\tau} - r_{f,\tau}) + \beta_{j,t}^{\mathrm{V}}dVIX_{\tau} + \beta_{j,t}^{\mathrm{Q}}dVVIX_{\tau} + \beta_{j,t}^{\zeta}\zeta_{\tau} + \epsilon_{j,\tau}.$$
 (15)

 ζ_{τ} contains one or more market factors, such as the Fama & French (1993) and Hou et al. (2015b) factors, the innovations in market skewness and kurtosis (dSkew, dKurt) as shown by Chang et al. (2013), the Cremers et al. (2015) straddle vol and jump factors, or innovations in the market variance risk premium (Han & Zhou, 2012; dVRP). For the regressions in Equations (14) and (15), we use daily returns over rolling annual periods to estimate the sensitivities. For each period and stock, we require at least one hundred non-missing return observations in order to estimate the factor sensitivities.⁹

Turning the focus on the measurement of innovations in an economic variable, there

⁹For the factor loading estimation regressions, potentially low explanatory power might be a concern. In the basic specification, we find the model in Equation (14) to exhibit an average R-squared of 0.31 with median 0.29. Thus, it can be concluded that the factor loading regressions possess substantial explanatory power. In these regressions, the coefficient $\beta_{j,t}^{\rm M}$ is significantly different from zero (at 10%) in 96% of the cases while the coefficients $\beta_{j,t}^{\rm Q}$ are significantly different from zero in about 19% and 13% of the cases, respectively.

generally exists a trade-off between a possible errors-in-variables problem using simple first differences, if that fails to completely filter out the expected movement, versus the danger of misspecifying a more complex equation for the expected movement in a variable (Chen et al., 1986). We choose to measure the innovations in the VVIX index using the daily first differences in the variable, similar to the way Ang et al. (2006b) obtain innovations in the VIX, because the VVIX is highly serially correlated with a first-order autocorrelation of 0.94 during our sample period. Therefore, the current value of VVIX appears to be a relatively good proxy for tomorrow's expectation making the first difference quite adequately capture its innovation. For robustness, we also consider measuring innovations in aggregate volatility-of-volatility by fitting an ARMA(1,1) model on the complete time series of the VVIX index. This approach results in a measure of innovations of $dVVIX_{\tau} = VVIX_{\tau} - 0.9989VVIX_{\tau-1} + 0.0839dVVIX_{\tau-1}$. The results of both approaches are qualitatively similar, which is further discussed in the next section.¹⁰

IV Empirical Results

A Descriptive Statistics

In addition to various firm characteristics, we consider the impact of several aggregate state variables that have previously been examined in the literature. In Table 1 we report the sample correlations between daily innovations in aggregate volatility-of-volatility (dVVIX), innovations in aggregate volatility (dVIX; Ang et al., 2006b), the Fama & French (1993), Carhart (1997), and Hou et al. (2015b) factors, and also the factors on market skewness and kurtosis of Chang et al. (2013), stochastic volatility and jump risk of Cremers et al. (2015), and innovations in the market variance risk premium (Han & Zhou, 2012).

First, we note that whether innovations are measured as simple first differences (dVVIX)

¹⁰One may argue that the innovation in the VVIX captures both the innovation in aggregate volatilityof-volatility and in the volatility-of-volatility risk premium. However, trying to estimate the risk premium as the difference between the expected volatility-of-volatility, obtained using VIX futures and the VVIX as described for the market variance risk premium in Appendix A, and the risk-neutral expectation, we find the innovations in the risk premium to be only weakly related to total innovations in the VVIX with a correlation of -0.12. Furthermore, when separating innovations in the VVIX from innovations in the risk premium and testing their significance in a joint cross-sectional regression, we find that only the "true" innovations in aggregate volatility-of-volatility are significantly priced. These results are available upon request.

or as innovations in an ARMA(1,1) model (dVVIX_{ARMA}) in fact does not make a difference. The two innovation measures are perfectly correlated. There is a negative correlation of dVVIX with MKT of -0.55. The linear relation between dVVIX and dVRP, jump, and dVIX, to which it might be related by construction, is of similar magnitude.¹¹ Correlations of dVVIX with other factors are negligible. Consequently, aggregate volatility-of-volatility appears to be distinct from other factors documented in the previous literature. In Table A1 of the online appendix, we also examine the Spearman rank correlations of the aggregate factor sensitivities. In general, these are very low. The rank correlation of aggregate volatility-ofvolatility and volatility sensitivities is even negative with -0.40 and substantially smaller in magnitude than the correlation of factor innovations.

We provide further summary statistics in Table 2. In Panel A it can be seen that mean and median innovations in the VVIX are close to zero. Measuring innovations in the VVIX using the first difference is shown to result in a factor with very low autocorrelation (-0.13), whereas using residuals from the fitted ARMA-model reduces the first-order autocorrelation to practically zero. The remaining factors are mostly constructed as returns with means close to zero and negligible autocorrelations.

Panels B and C of Table 2 present yearly summary statistics on the VVIX and $\beta_{j,t}^{Q}$ factors of individual stocks, respectively. It is quite interesting to observe that the yearly average level of the VVIX is smallest in the crisis year 2009. In the years 2011, 2012, 2015, and 2016 the average volatility-of-volatility is substantially higher with values above 0.9. In 2013, there is a sharp decrease, almost returning to the 2009 level. The mean and median sensitivities of individual stocks to innovations in aggregate volatility-of-volatility, $\beta_{j,t}^{Q}$, presented in Panel C, are mostly close to zero. The highest average $\beta_{j,t}^{Q}$ and average cross-sectional standard deviation results for rolling annual estimation periods starting in 2008.

The time series of the VIX and VVIX are plotted in Figure 1. We find that the level of the VVIX is higher than that of the VIX throughout the entire sample period. The averages of the VVIX and VIX are 0.87 and 0.20, respectively. Both, the S&P 500 and the VIX

¹¹Bollerslev et al. (2009) show that volatility-of-volatility is theoretically (positively) linked to the variance risk premium. However, the sign depends on the definition of the variance risk premium as risk-neutral minus physical expected variance. While we define it the other way around, our results are fully consistent with their derivation.

are directly tradable, e.g., using futures on the respective indexes. Thus, the underlyings of the option-implied volatilities are comparable and our results indicate that the risk-neutral expectations of the variance of the returns on the VIX is substantially higher than that of the returns on the S&P 500. Furthermore, the VVIX itself is by far more volatile and less persistent compared to the VIX. The VVIX exhibits pronounced spikes that correspond with news increasing economic uncertainty. For these news, typically the spikes in the VIX are by far less pronounced or not perceptible. These news events include, e.g., the Bear Sterns Hedge Funds Collapse (August 2007), the Lehman Brothers bankruptcy (September 2008), the Freddie Mac and Fannie Mae crisis (May 2010), and the near collapse of the Russian ruble (December 2014). Consequently, this stylized evidence provides further insights for the notion that VVIX is a proxy for economic uncertainty, distinct from volatility.

One may wonder whether the VVIX helps predict future volatility and volatility-ofvolatility. To analyze this, we estimate predictive regressions for the realized volatility of the S&P 500 and the VIX. We use the end-of-month observations of the VIX and VVIX to predict the realized volatility, based on 5-minute returns, over the subsequent month. We present the results in Table 3. We find that, univariately, both the VIX and the VVIX significantly predict future realized variance. However, in a multivariate predictive regression, only the VIX significantly predicts the future realized volatility of the market index. On the other hand, we find that only the VVIX positively predicts the future realized volatility-ofvolatility, both in a univariate and a multivariate framework. These results are consistent with those in Huang & Shaliastovich (2014) for a shorter sample period.

Overall, the results indicate that aggregate volatility-of-volatility, measured by the VVIX, and aggregate volatility, measured by the VIX, have differential dynamics and are only weakly related.

B Portfolio Sorts and Characteristics

To test whether aggregate volatility-of-volatility is priced in the stock market, we first perform portfolio sorts. At the beginning of each month, we sort the stocks in ascending order with respect to their sensitivities to innovations in volatility-of-volatility ($\beta_{j,t}^{Q}$) over the following year. We form quintile portfolios, so that quintile 1 contains the stocks with the lowest exposure to aggregate volatility-of-volatility while quintile 5 contains those stocks with the highest volatility-of-volatility factor loadings. The hedge portfolio (5 minus 1) buys the quintile of stocks with the highest exposure and simultaneously sells the stocks in the quintile with the lowest exposure to aggregate volatility-of-volatility. The portfolio sorting approach maximizes the spread in the exposure to aggregate volatility-of-volatility and, thus, differences in average returns can be quite accurately attributed to differences in the sorting variable. Fama & French (2008) point out that by building value-weighted portfolios the hedge portfolio can be dominated by few big stocks, whereas for equally weighted portfolios the hedge portfolio can be dominated by micro caps. To address this issue, we analyze both value-weighted and equally weighted portfolio sorts. For the portfolio sorts, within each quintile, we weight the stocks by their relative market value at the beginning of the estimation period for $\beta_{j,t}^{Q}$. While our research design involves successive twelve-month periods employing partly overlapping information, it introduces moving average effects. To account for that, in all analyses, we adjust the standard errors following Newey & West (1987) using twelve lags.¹²

Table 4 reports various summary statistics for the quintile portfolios sorted by contemporaneous volatility-of-volatility-sensitivities. We find that the average annual raw return obeys a nearly strictly monotonically decreasing pattern, from 10.0% and 10.6% in quintiles 1 and 2 to 5.1% in quintile 5. Consequently, the better the stocks insure against positive shocks to aggregate volatility-of-volatility, the lower are their returns on average. The difference in raw returns of -4.91% between quintiles 5 and 1 is statistically significant with a p-value of 2.7%. Looking at the line labeled CAPM alpha, which reports the results when controlling for systematic risk, we find a similar abnormal return for the 5 minus 1 portfolio of -4.87%, which is also significant at 5%. Controlling for the Carhart (1997) factors (4-factor) and additionally including the Pastor & Stambaugh (2003) factor (5-factor) leads to alphas of -4.91% and -4.62% per year, both significant at 5%. The alpha relative to the new factor model by Hou et al. (2015b) amounts to -6.35%, which is also significant at 5%.¹³ Conse-

 $^{^{12}}$ While theoretically only eleven lags are required, we follow Ang et al. (2006a) and Cremers et al. (2015) including an additional lag for robustness.

¹³In light of the results of Fabozzi et al. (2016), we decide to use the model of Hou et al. (2015b) instead of that of Fama & French (2015). We thank the editor, Raman Uppal, for suggesting this.

quently, accounting for systematic risk factors, the 5 minus 1 portfolio is expected to earn substantially higher returns than realized. This effect is mostly driven by the portfolio of the stocks with the highest sensitivities to innovations in aggregate volatility-of-volatility. For each factor model specification, the alpha of this portfolio is strongly significantly negative, while that of portfolio 1 is close to zero. Hence, investors are willing to pay a premium in order to hold stocks with high sensitivities to innovations in aggregate volatility-of-volatility while the stocks with the lowest sensitivities do not earn significant positive abnormal returns. We obtain similar results for value-weighted returns. The return and the alpha on the 5 minus 1 portfolio is typically even larger than for the equally weighted approach, although the significance of the abnormal returns of the hedge portfolio is typically slightly weaker.

In the remainder of Table 4, we examine value-weighted characteristics of the portfolios. The 5 minus 1 portfolio has a significantly positive exposure to the market excess return. The average market beta of the hedge portfolio of 0.31 would, following the logic of the CAPM, predict a substantially positive excess return, whereas the return that is realized is significantly negative. Interestingly, the hedge portfolio has a significantly negative exposure to aggregate volatility. The exposure to aggregate volatility-of-volatility is monotonically increasing by construction. Furthermore, we find that the firms in portfolio 5 on average are smaller than those in portfolio 1, have higher book-to-market ratios, are younger, and have slightly higher bid-ask spreads and leverage.

Another interesting issue relates to the time-series of excess returns of the 5 minus 1 hedge portfolio, presented in Figure 2. First, this analysis shows whether the cumulative returns are generated by a certain time period. Second, the plot also shows the periods where the insurance strategy pays off. To obtain a time-series of monthly returns, we average the returns of all active portfolios during that month.¹⁴ We present the results for both value- and equally weighted portfolios in Figure 2. Over the entire sample period, the cumulative excess return is roughly -37% for the value-weighted and -32% for the equally weighted portfolios. The trend is in general negative but there are also periods for which the insurance strategy pays off and the 5 minus 1 portfolio yields large positive returns. These times roughly

 $^{^{14}\}mathrm{Due}$ to the overlapping approach with 12-month holding periods, typically, there are 12 active 5 minus 1 hedge portfolios.

correspond with the first half of 2008, where the National Bureau of Economic Research (NBER) indicates a recession, as well as short time periods in the beginning of 2009, in the beginning of 2012, in the second half of 2014, and in the beginning of 2016. These time periods typically correspond with positive innovations in and correspondingly high levels of the VVIX index.

C Double Sorts

Given the close relation of aggregate volatility-of-volatility and aggregate volatility and also the market variance risk premium (as shown by Bollerslev et al., 2009), we aim to further dissect the effects of these variables. To do that, we estimate the factor sensitivities in a joint regression and perform a double-sort.¹⁵

At the beginning of each month we independently sort all stocks into quintiles, using NYSE breakpoints, on the characteristic we want to control for as well as on their VVIX-sensitivities (Novy-Marx, 2013).¹⁶ This results in a total of 25 portfolios. Thus, we obtain quintile portfolios on the exposure to aggregate volatility-of-volatility controlling for another characteristic without making assumptions on the parametric form of the relationships. For each quintile of both sorting variables, we obtain a 5 minus 1 hedge portfolio by buying the final portfolio 5 and selling portfolio 1 in the quintile. We consider both equally and value-weighted portfolios. For value-weighted portfolios, within each of the 25 portfolios we weight the stocks by their relative market value at the beginning of the estimation period for the exposures to aggregate volatility-of-volatility.

Table 5 reports all 25 portfolios resulting from the double sorts as well as the respective 5 minus 1 portfolios. We report Hou et al. (2015b) factor alphas and robust Newey & West (1987) p-values in brackets.¹⁷ Panels A and B report the results for equally and value-weighted double sorts with respect to volatility-of-volatility and volatility betas. According

¹⁵For double sorts with aggregate volatility, we directly obtain the betas from Equation (14). When controlling for the variance risk premium, we use the same regression equation, replacing $dVIX_{\tau}$ by $dVRP_{\tau}$.

¹⁶While the double-sorting approach we use is unconditional, the results are qualitatively similar when performing a conditional double sort, i.e., first sorting on a certain characteristic and subsequently sorting on the VVIX-sensitivities. These results are available upon request.

¹⁷The results for Carhart (1997) 4-factor alphas are qualitatively similar. These results are available upon request.

to the model's result in Equation (13), there are three risk factors in the market. Stock returns depend on the stocks' $\beta_{j,t}^{M}$, $\beta_{j,t}^{V}$, and $\beta_{j,t}^{Q}$. Hence, given that the price of risk on both volatility and volatility-of-volatility appear to be negative, stocks with low $\beta_{j,t}^{V}$ and low $\beta_{j,t}^{Q}$ should earn high average returns. For each quintile of $\beta_{j,t}^{V}$, stocks with higher $\beta_{j,t}^{Q}$ should earn lower returns. Similarly, for each quintile of $\beta_{j,t}^{\mathbf{Q}}$, stocks with higher $\beta_{j,t}^{\mathbf{V}}$ should earn lower returns. This is exactly the pattern we find in Panels A and B of Table 5. Portfolios with both low $\beta_{j,t}^{V}$ and low $\beta_{j,t}^{Q}$ have positive alphas. Both, with increasing volatility-betas and volatility-of-volatility betas, the alphas decrease. The largest negative alpha results for the portfolio of stocks with the highest values for both $\beta_{j,t}^{V}$ and $\beta_{j,t}^{Q}$. For all volatility-of-volatility quintiles, we detect significantly negative alphas for the 5 minus 1 portfolios. The alphas range between -13.7% and -12.0% for equally weighted returns and are even larger in magnitude for value-weighted returns. Similarly, for all volatility quintiles, beside quintile 5 for value-weighted returns, we find significantly negative alphas for the 5 minus 1 portfolios. The alphas range between -11.0% and -8.0% for equally weighted returns and -12.2% and -6.4% for value-weighted returns. Consequently, as also indicated by the negative loading of the 5 minus 1 portfolio on aggregate volatility in Table 4, when controlling for aggregate volatility, the trade-off of volatility-of-volatility and returns is even more pronounced than for the univariate sorts.

In Panels C and D of Table 5, we report the results of double sorts of volatility-ofvolatility betas and betas with respect to the variance risk premium. For all quintiles of volatility-of-volatility betas, we find positive 5 minus 1 portfolio alphas on the variance risk premium. And for all variance risk premium beta quintiles, we detect a significantly negative hedge portfolio alpha for volatility-of-volatility. Bollerslev et al. (2009) find that the variance risk premium is associated with macroeconomic volatility-of-volatility. In light of these results, one may argue that our result that both aggregate volatility-of-volatility and the market variance risk premium are priced is surprising. However, estimating the variance risk premium faces the problem of specifying an expectation of the variance under \mathbb{P} . Empirically, innovations in the market variance risk premium are mostly driven by innovations in the risk-neutral expectation. dVIX and dVRP are highly correlated with -77%. Hence, empirically, innovations in the market variance risk premium mostly capture innovations in aggregate volatility rather than consumption volatility-of-volatility. This suggests that it is important to directly use stock market volatility-of-volatility instead of simply following Bali & Zhou (2015) in proxying consumption volatility-of-volatility with the market variance risk premium.

D Regression Tests

The portfolio sorts present strong evidence that sensitivities to innovations in aggregate volatility-of-volatility are related to stock returns. The double sorts indicate that this relation persists when controlling for, and hence cannot be explained by, aggregate volatility or the variance risk premium. Following on from that, in this section, we estimate Fama & MacBeth (1973) regressions that simultaneously control for different variables and test if a stock's sensitivity to innovations in aggregate volatility-of-volatility contains information about stock returns beyond that of various other firm characteristics and factor sensitivities. We perform the regressions using individual stocks instead of stock portfolios. Lo & MacKinlay (1990) and Lewellen et al. (2010) argue against the use of portfolios in cross-sectional regressions, since the particular method, by which the portfolios are formed can severely influence the results. Furthermore, Ang et al. (2010) show that creating portfolios ignores important information on individual factor loadings and leads to higher asymptotic standard errors of risk premium estimates. Consequently, using individual stocks, we utilize this additional information and, at the same time, avoid the specification of breakpoints.

Each month, we perform cross-sectional regressions of stock excess returns over the following year on the stocks' sensitivities to innovations in aggregate volatility-of-volatility and several control variables, measured over the same period. We winsorize all regressors at the 1st and 99th percentile to restrict the effect of outliers (Fama & French, 2008; Baltussen et al., 2015).¹⁸ For the regressions, we use ordinary least squares (equally weighted) or weighted least squares (value-weighted) with a diagonal weighting matrix, where the inverse of the firm's market value at the beginning of the estimation period for $\beta_{j,t}^{Q}$ is along the

¹⁸The results of non-winsorized regression tests are qualitatively similar. These results are available upon request.

diagonal, with the following regression equation:

$$r_{j,t} - r_{f,t} = \lambda_t^0 + \lambda_t^M \beta_{j,t}^M + \lambda_t^V \beta_{j,t}^V + \lambda_t^Q \beta_{j,t}^Q + \lambda_t^\zeta \beta_{j,t}^\zeta + \epsilon_{j,t}.$$
 (16)

 $r_{j,t}$ is the annual return of stock j and $r_{f,t}$ is the risk-free rate during that period. $\beta_{j,t}^{\mathrm{M}}$, $\beta_{j,t}^{\mathrm{V}}$, and $\beta_{j,t}^{\mathrm{Q}}$ are the stock's market beta, sensitivity to innovations in aggregate volatility and sensitivity to innovations in aggregate volatility-of-volatility over the evaluation period, respectively. The term $\beta_{j,t}^{\zeta}$ denotes a vector collecting further variables hypothesized to explain returns. λ_t^0 denotes the regression intercept and λ_t^{M} , λ_t^{V} , λ_t^{Q} , and λ_t^{ζ} are the prices of factor risk associated with the respective variables. $\epsilon_{j,t}$ is the idiosyncratic return component of stock j at time t.

In the next step, we perform tests on the time series averages $\bar{\lambda^0}$, $\bar{\lambda^M}$, $\bar{\lambda^V}$, $\bar{\lambda^Q}$, and $\bar{\lambda^\zeta}$ of the estimated intercept and slope coefficients, $\hat{\lambda^0_t}$, $\hat{\lambda^V_t}$, $\hat{\lambda^Q_t}$, and $\hat{\lambda^\zeta_t}$. We account for potential autocorrelation, heteroskedasticity, and errors-in-variables concerns, computing robust Newey & West (1987) (again using twelve lags) and Shanken (1992) adjusted p-values based on the time series of coefficient estimates.

Table 6 reports the results of the basic Fama & MacBeth (1973) regressions. We report the results of a regression of excess returns on $\beta_{j,t}^{M}$, $\beta_{j,t}^{V}$, $\beta_{j,t}^{Q}$, and various other canonical characteristics. In the basic regression specification suggested by our theoretical model (i), the price of risk on the market is essentially zero and the price of risk on aggregate volatility amounts to -0.2279, which is significant at 5%. The annual factor risk premium of aggregate volatility-of-volatility ($\overline{\lambda^{Q}}$) is -0.7753 with a p-value of 0.010.¹⁹ Consequently, a two-standard deviation increase in a stock's volatility-of-volatility-sensitivity is associated with a 10.61% decrease in average annual returns.²⁰

Adding further control variables, i.e., size, book-to-market, bid-ask spread, momentum, short-term reversal, age, and leverage in models (ii) to (viii) does not change our main results. The risk premium estimate on volatility-of-volatility remains economically large with values

¹⁹Note that the correlation between factor loadings on dVVIX and dVIX amounts to only -40% (compared to the factor correlation of 67%), which makes it unlikely that problems due to multicollinearity are present.

²⁰This number is obtained as follows. We multiply the risk premium estimate with two times the average cross-sectional standard deviation of the sensitivities to innovations in aggregate volatility-of-volatility from Table 2 (0.0684). Plugging in yields -0.7735 * (2 * 0.0684) = -0.1061.

ranging from -0.81 to -0.70 and highly significant for every specification.

Adding all canonical characteristics jointly yields a factor risk premium of aggregate volatility-of-volatility of -0.69, which is statistically significant at 5%. The risk premium estimate on the market excess return is not significantly different from zero and even negative for some specifications. This result is consistent with recent evidence that market beta is not priced in the cross-section of stock returns (e.g., Frazzini & Pedersen, 2014). Many of our control variables yield insignificant risk premium estimates for our sample period or conditional risk premium estimates with different signs compared to the previous literature. Hence, our results are in line with the view that prominent return anomalies have attenuated recently (Chordia et al., 2014). Adding characteristics univariately, we find a positive conditional risk premium on size and negative conditional risk premia on book-to-market and the bid-ask spread. Hence, stocks of large firms and stocks with lower bid-ask spreads, conditional on all remaining variables being the same, earn higher returns than stocks of small firms and stocks with higher bid-ask spreads and value stocks earn lower returns than growth stocks during the recent sample period. On the other hand, momentum, short-term reversal, age, and leverage do not obtain significant prices of risk, conditional on our main model specification.

Table A2 of the online appendix repeats the analysis for value-weighted regressions. The results are qualitatively similar. We obtain risk premium estimates on aggregate volatility-of-volatility of around -0.8, which are statistically significant at 5% throughout. Additionally, in Table A3 of the online appendix, we present the results when controlling for further returns distributions and liquidity-related characteristics, like idiosyncratic volatility, co-skewness, or Amihud illiquidity. For each of the model specifications, we obtain a significantly negative cross-sectional price of risk on aggregate volatility-of-volatility.

Lastly, we examine whether the cross-sectional regression results are consistent with the model predictions of Equation (13). The first model prediction is that the cross-sectional risk premium on the market portfolio is positive and proportional to the coefficient of relative risk-aversion. For example, the coefficient of relative risk aversion in the basic model specification

(i) of Table 6 is approximately equal to zero.²¹ The coefficient of relative risk aversion therefore is not really in a reasonable range (Friend & Blume, 1975). For some specifications, the risk premium estimate on the market is even negative, implying a negative relative risk aversion. However, the standard error of the risk premium estimate on the market is typically large, e.g., it amounts to 6.26% in the basic model specification (i). Hence, the point estimate is ambiguous and has to be interpreted with caution. Next, we turn the focus on the price of risk on aggregate volatility, which is significantly negative throughout all test specifications. This is in line with the model's prediction. Finally, we find that the market price of risk on volatility-of-volatility, which could be positive or negative according to our theoretical model, is significantly negative. This result is consistent with Bali & Zhou (2015), who proxy consumption volatility-of-volatility with the variance risk premium.

V Further Analyses

A Multivariate Estimation

In this section, we examine the robustness of our results to jointly estimating the sensitivities to innovations in aggregate volatility-of-volatility with those to other factors, as presented in Equation (15). Table 7 reports the results of Fama & MacBeth (1973) regressions using multivariate sensitivity estimation regressions. Incorporating the Fama & French (1993) factors in model (i) leaves the risk premium estimate of aggregate volatilityof-volatility strongly significant at 1%. Adding the other market factors like dSkew, dKurt, straddle vol, jump, or dVRP (models (ii) to (vii)) only marginally affects the risk premium estimate on aggregate volatility-of-volatility. It ranges from -0.76 to -0.71 and remains highly statistically significant. Adding the Hou et al. (2015b) factors in a joint sensitivity estimation, the risk premium estimate on aggregate volatility-of-volatility gets even more negative, reaching a value of -0.90 and is statistically significant at 1%.

The price of risk on HML regularly is significantly negative, while that on MKT and SMB is never significantly different from zero. The prices of risk on jump is significantly negative

 $^{^{21}}$ The coefficient of relative risk aversion of the representative investor can be approximated by dividing the cross-sectional price of risk by the variance of the market return, which amounts to 0.0019.

and those on straddle vol and dVRP are significantly positive. Regarding the new investment and profitability factors of Hou et al. (2015b), ROE obtains a positive risk premium estimate while that on IA is indistinguishable from zero. The results for value-weighted regressions, presented in Table A4 of the online appendix, are qualitatively similar.

B Orthogonalized Factor Innovations

To account for concerns related to high correlations among market excess returns, innovations in the VIX, and innovations in the VVIX, this section reports results for orthogonalized factor innovations. We perform the orthogonalization following Sims (1980). We orthogonalize innovations in the VIX with respect to the market excess return and innovations in the VVIX with respect to both the market excess return and the VIX. With this order of orthogonalization, it is least likely that we find a positive market price of risk on aggregate volatility-of-volatility, since the common variation with the market excess return and aggregate volatility is removed from the variable.

The orthogonalization approach delivers two important insights. First, we can directly assess the market prices of risk on the factors which potentially deviate from the factor risk premia of a cross-sectional regression when the factors are strongly correlated. Second, we can test whether there is a clear incremental contribution of the information contained in aggregate volatility-of-volatility in explaining the cross-sectional variation in equity returns compared to only using the information contained in aggregate volatility and the market excess return.

The results are presented in Table 8. We find that, consistent with the results presented in Table 6, the market price of risk on the market excess return is not significantly different from zero. The market price of risk on aggregate volatility is statistically significant at 1% in any case and is slightly smaller in magnitude compared to the non-orthogonalized version of Table 6, with values ranging between -0.22 and -0.16. Hence, the information contained in aggregate volatility has a strong marginal contribution in explaining the crosssection of stock returns relative to the market excess return. Finally, the market price of risk on the orthogonalized innovations in aggregate volatility-of-volatility ranges between -0.41and -0.35. The market price of risk of the orthogonalized factor innovations is only about half that of the non-orthogonalized innovations. Hence, as indicated by the correlation of innovations in volatility-of-volatility with those in volatility as well as the market excess return, there seems to be some interaction of these factors. However, the market price of risk on the orthogonalized aggregate volatility-of-volatility is still significant with p-values ranging between 3.7% and 8.8%. For model specification (i), a two standard deviation increase in factor loadings on orthogonalized volatility-of-volatility innovations is associated with a decrease in annual returns of about 5.2%. Consequently, we find compelling evidence that volatility-of-volatility is a priced risk factor in the cross-section of equity returns, over and above aggregate volatility.

C Predicting Future Factor Loadings

In Section IV, we demonstrate a strong negative relation between the contemporaneous sensitivities to innovations in aggregate volatility-of-volatility and stock returns. While the previous analysis strongly indicates that volatility-of-volatility is a priced risk factor, for practical purposes it is also of interest how this risk can be hedged. Therefore, in this section, we first examine whether one can create the same return spread using pre-formation factor loadings. In the second part of this section, we examine which past sensitivities and stock characteristics best predict future volatility-of-volatility sensitivities.

We start by examining the relation of average returns and past sensitivities to innovations in aggregate volatility-of-volatility. Ang et al. (2006a) point out that pre-formation factor loadings, beside actual variation in exposures toward that factor, additionally reflect measurement error. This measurement error is likely to be especially high during highly volatile times, like, in particular, in the first part of our sample period. Taken together, noisy estimates of our sorting variable and a relatively short sample increase the standard error estimates and yield a test with reduced statistical power to detect significant spreads in returns and alphas. Hence, while for these reasons we concentrate on the results generated using post-formation factor loadings, the results from pre-formation factor loadings might be weaker than those presented in the previous sections.

Table 9 reports these results. Sorting the stocks by past exposure to aggregate volatilityof-volatility fails to produce any spread in average returns and Hou et al. (2015b) factor alphas. The main explanation for this is given by the realized ex post factor sensitivities of the portfolios. In the presence of higher measurement errors, the spread in ex post exposures to aggregate volatility-of-volatility is only 0.002 for equally weighted portfolios and even negative with -0.006 for value-weighted portfolios. This spread is substantially smaller than the 0.13 produced by sorting on contemporaneous factor loadings (see Table 4). Hence, the current factor loadings on average have very weak predictability for future factor loadings in aggregate volatility-of-volatility.

Given the weak results when using pre-formation factor loadings, we examine the crosssectional predictability of several ex ante firm characteristics and factor sensitivities for future sensitivities to innovations in aggregate volatility-of-volatility by performing Fama & MacBeth (1973) regressions. Cross-sectional predictors of the stocks' volatility-of-volatilitysensitivities are measured during the twelve months directly prior to the twelve-month estimation period for sensitivities to innovations in aggregate volatility-of-volatility. The results are reported in Table 10. Consistent with the results for the pre-formation factor loadings in Table 9, we find that past volatility-of-volatility-sensitivities do not significantly predict future relative volatility-of-volatility-sensitivities.²² Hence, there appears to be large timevariation in factor loadings on aggregate volatility-of-volatility. The best predictors for future volatility-of-volatility factor sensitivities are past beta with a positive sign and the bid-askspread, momentum, short-term reversal, idiosyncratic volatility, and downside beta with a negative sign. The adjusted \mathbb{R}^2 , however, is only 8.3% when including all predictors. Hence, although we find some characteristics that are related to future volatility-of-volatility factor loadings, it is difficult to predict future factor loadings and hedge volatility-of-volatility risk in the stock market.²³

D Hedging Volatility-of-Volatility Risk

The preceeding sections show that volatility-of-volatility is significantly priced in the stock market but difficult to hedge. In this section, we examine the VIX options market that likely

 $^{^{22}}$ We find that the factor loadings on the VIX are slightly better predictable than those on the VVIX, but not very much. The average 12-month autocorrelation of VIX factor loadings 7.4%.

²³We experimented with various possibilities to predict future volatility-of-volatility factor loadings. However, we could not create a significant spread in returns using these predictors.

delivers better hedging opportunities.²⁴ Since the vega, i.e., the sensitivity of the options price with respect to the volatility of the unerlying, of all options is positive, these are natural instruments to hedge against increases in the volatility of the underlying. We follow Bakshi & Kapadia (2003) and Huang & Shaliastovich (2014), computing continuously-rebalanced delta-hedged option returns. The delta-hedged VIX-option position is approximately insensitive to innovations in aggregate volatility. Hence, it only delivers insurance against increases in aggregate volatility. The option gain can be computed as follows (Huang & Shaliastovich, 2014):

$$\Pi_{t,T} = M_T - M_t - \sum_{\tau=0}^{N-1} \Delta_{t_\tau} \left(F_{\tau+1} - F_{\tau} \right) - \sum_{\tau=0}^{N-1} r_{f,\tau} M_t \frac{T-t}{N}.$$
(17)

 $\Pi_{t,T}$ is the option gain and F_t is the forward price of the option's underlying. We divide $\Pi_{t,T}$ by the forward price F_t to obtain the delta-hedged return on the option. M_T is the final payoff of the option (put or call) at time T and M_t is the option price at initiation t. $\Delta_{t_{\tau}}$ is the Black & Scholes (1973) delta of the option at time τ . N is the number of days between t and T that are trading days. All other variables are as previously defined. We use all options with volume and open interest greater than zero. For the strategy, we buy options with 30 days to maturity and hold them until expiration, rebalancing the delta-hedge at the end of each trading day.

We present the results for delta-hedged option returns in Table 11. Consistent with Huang & Shaliastovich (2014), who end their sample period in September 2012, we find negative delta-hedged option returns both for calls and puts for each moneyness bin, with moneyness computed as strike price over futures price at initiation $\left(\frac{K}{F_{t}}\right)$, between 0.7 and 1.3. These returns are significantly negative for in-the-money calls and both in-the-money and out-of-the-money puts. This finding confirms our previous results of a negative risk premium on volatility-of-volatility. On average, investors on the options market are also willing to pay a premium in order to insure against increases in aggregate volatility-of-volatility. Worth noting are also the distributional characteristics of the delta-hedged option returns. The returns are highly positively skewed with skewness of 0.74 up to 1.68 and have high kurtosis

²⁴We thank an anonymous referee for suggesting this.

between 8.36 and 9.13. The 5% quantile of option return lies between -4.8% and -2.0% for the different moneyness bins. The 95% quantile lies between 1.3% and 7.8%. Hence, the option return can become strongly positive. These distributional characteristics further indicate that the options on the VIX are adequate hedges for volatility-of-volatility risk.²⁵

E Realized Measure of Aggregate Volatility-of-Volatility

Taking account of the literature on volatility estimation, we further test the robustness of our results employing an inferior measure of aggregate volatility-of-volatility. Instead of utilizing the forward-looking expected volatility-of-volatility, in this section, we rely on the realized volatility of the VIX. To perform the analysis, we use five-minute high-frequency data on the VIX from the TRTH database and estimate each day the realized bipower volatility over the past month (Barndorff-Nielsen & Shephard, 2004). Since the high-frequency data is available for a longer period, we extend the sample period to January 01, 2000 until September 30, 2016. We perform standard data cleaning operations following Barndorff-Nielsen et al. (2009). Following on from that, we obtain daily realized volatilities of the VIX and compute the innovations $(dVoVIX_t)$ again as simple first differences. We estimate factor sensitivities using Equation (14), replacing dVVIX by dVoVIX.

We present the results of portfolio sorts based on dVoVIX in Table A5 of the online appendix. The return differential between stocks with high and those with low sensitivities to innovations in aggregate volatility-of-volatility is only slightly smaller than when using the VVIX in Table 4 with about -3.4%. The return differential is not significant with a p-value of 0.19. However, once controlling for systematic risk, we find a significant volatilityof-volatility-return trade-off with alphas relative to the CAPM, 4-factor, and 5-factor models of -6.6% up to -4.4% which are all significant. When using the Hou et al. (2015b) factor model, the alpha becomes insignificant. When restricting the sample to the horizon in which data on the VVIX is available (2006 until 2016), we also do not find a significant return differential but a significant Hou et al. (2015b) alpha.

Even though the portfolio of stocks with high sensitivities to innovations in aggregate

 $^{^{25}\}mathrm{Huang}$ & Shaliastovich (2014) further show that volatility-of-volatility can predict future delta-hedged option returns.

volatility-of-volatility has highly significantly negative alphas in most cases, these are not directly carried over to the hedge portfolio since the alpha of the portfolio with the lowest sensitivities is close to zero or also slightly negative on average. Hence, overall the model prediction that investors strongly demand stocks which insure against increases in aggregate volatility-of-volatility is confirmed by these results. However, using a realized measure of volatility-of-volatility, the monotonicity in portfolios 1 to 4 vanishes. Consequently, when using this theoretically inferior measure, there is still some indication of a volatility-of-volatility-of-volatility-of-the second structure is still some indication. Therefore, it is important to use the forward-looking information from options markets.²⁶

VI Conclusion

Using a simple stylized theoretical model with Epstein–Zin preferences and consumption uncertainty, we motivate a potential trade-off of volatility-of-volatility and stock returns. In our empirical study, we confirm this model prediction. We find that aggregate volatility-ofvolatility carries a considerable and robust negative risk premium.

We use both uni- and bivariate portfolio sorts and show that the quintile portfolio of stocks with the highest sensitivity toward innovations in aggregate volatility-of-volatility underperforms the quintile of stocks with the lowest exposure to aggregate volatility-ofvolatility by about 5% per annum. Using regression tests, we estimate the cross-sectional price of factor risk of aggregate volatility-of-volatility to be both economically substantial and statistically highly significant. The estimated factor price on aggregate volatility-ofvolatility cannot be explained by known risk factors. Our results are also consistent with any multi-factor model, in which aggregate volatility-of-volatility is priced with a negative sign if investors relate a positive change in aggregate volatility-of-volatility to future unfavorable shifts in the investment opportunity set.

 $^{^{26}}$ This result is similar to that of Ang et al. (2006b), who only find a significant spread in the cross-section of returns when using an option-implied, as opposed to a simple historical or high-frequency, measure to proxy for innovations in aggregate volatility.

Appendix

A Variable Definitions – Main Control Variables

- Age (Zhang, 2006) is the number of years up to time t since a firm first appeared in the CRSP database. In regressions, we take the natural logarithm to remove the extreme skewness in this variable.
- Aggregate volatility (Ang et al., 2006b, "dVIX"), is the coefficient $\beta_{j,t}^V$ obtained by the regression in Equation (14).
- **Beta** is the coefficient $\beta_{j,t}^M$ obtained by the regression in Equation (14).
- **Bid-ask spread** is the stock's average daily relative bid-ask spread over the examination period.
- Book-to-market (Fama & French, 1992) is the weighted average of book equity divided by market equity over the examination period. Book and market equity are updated every 12 months for the beginning of the year. Book equity is defined as stockholders' equity, plus balance sheet deferred taxes and investment tax credit, plus post-retirement benefit liabilities, minus the book value of preferred stock.
- Leverage (Bhandari, 1988) is defined as the weighted average of one minus book equity (see "Book-to-market") divided by total assets (Compustat: AT). Book equity and total assets are updated every 12 months for the beginning of the year.
- Market factors: "MKT" is the value-weighted excess return on all CRSP firms. "SMB", and "HML" are the Fama & French (1993) size and value factors. "SMB^{HXZ}", "IA", and "ROE" are the size, investment, and profitability factors of Hou et al. (2015b).
- Market variance risk premium (Han & Zhou, 2012, "dVRP"), is the coefficient $\beta_{j,t}^{dVRP}$ in the regression $r_{j,\tau} r_{f,\tau} = \alpha_{j,t} + \beta_{j,t}^M (r_{M,\tau} r_{f,\tau}) + \beta_{j,t}^{dVRP} dVRP_{\tau} + \epsilon_{j,\tau}$, using daily returns over the examination period (Ang et al., 2006a), where the market variance risk premium is defined as the difference between the physical expected variance

and the risk-neutral expected variance (VIX^2) of the S&P 500 index over a 30-day horizon using daily return data. First, we compute the expected variance (EV_{τ}) under the physical measure by regressing the annualized realized variance $(RV_{\tau+30})$ on the lagged implied (VIX_{τ}^2) and the lagged annualized historical (RV_{τ}) realized variance, using an expanding window of daily data that is available at time τ , starting with data from January 01, 1996 ($\tilde{\tau}$ refers to those dates) $RV_{\tau+30} = \alpha_{\tau} + \beta_{\tau}VIX_{\tau}^2 + \gamma_{\tau}RV_{\tau} + \epsilon_{\tau+30}$ in a first step and then computing $EV_{\tau} = \hat{\alpha}_{\tau} + \hat{\beta}_{\tau}VIX_{\tau}^2 + \hat{\gamma}_{\tau}RV_{\tau}$. The market variance risk premium (VRP_{τ}) is obtained as $VRP_{\tau} = EV_{\tau} - VIX_{\tau}^2$. dVRP is obtained as the first difference in VRP.

- Momentum (Jegadeesh & Titman, 1993) is the cumulative stock return over the period from t 12 until t 1.
- Short-term reversal (Jegadeesh, 1990) is the preceding month's stock return (from t-1 to t).
- Size (Banz, 1981) is the average of firm's market capitalization over the examination period. Market capitalization is computed as the product of the price times the number of shares outstanding. In regressions, we take the natural logarithm to remove the extreme skewness in this variable.
- Stochastic volatility (Cremers et al., 2015, "straddle vol"), market skewness and kurtosis (Chang et al., 2013, "dSkew", "dKurt"), and aggregate jump risk (Cremers et al., 2015, "jump") are the coefficients $\beta_{j,t}^F$ in the regression $r_{j,\tau} - r_{f,\tau} = \alpha_{j,t} + \beta_{j,t}^M (r_{M,\tau} - r_{f,\tau}) + \beta_{j,t}^F F_{\tau} + \epsilon_{j,\tau}$, using daily returns over the examination period (Ang et al., 2006a), where F is one of the following:
 - Innovations in implied market skewness and kurtosis (Chang et al., 2013), which are defined as the difference of daily implied skewness (kurtosis) computed from S&P 500 index options using the formulas of Bakshi et al. (2003) and its expectation, which is obtained by fitting an ARMA(1,1) model on the complete time series of skewness (kurtosis) estimates. The resulting measure of innovations in market skewness then is $dSkew_{\tau} = Skew_{\tau} - 0.9993Skew_{\tau-1} + 0.6647dSkew_{\tau-1}$

and that of innovations in market kurtosis is $dKurt_{\tau} = Kurt_{\tau} - 0.9999Kurt_{\tau-1} + 0.4971 dKurt_{\tau-1}$.

Market-neutral straddle returns (Cremers et al., 2015) are computed by first constructing at-the-money zero beta straddles. Afterwards, the straddle vol factor is the return of a gamma neutral and vega positive portfolio of the two straddles maturing in the next month and the month after next, while the jump factor is the return of a gamma positive and vega neutral portfolio using the same straddles. To construct the factors, we use Black & Scholes (1973) option sensitivities.

B Variable Definitions – Further Controls

- Amihud illiquidity (Amihud, 2002) is the absolute value of the stock's return divided by the daily dollar volume, averaged over the examination period. Specifically, it is $Illiq_t = \frac{1}{n} \sum_{\tau=1}^n \frac{|r_{j,\tau}|}{Volume\$_{\tau}}$, with the daily dollar volume ($Volume\$_{\tau}$, in thousand dollars) being calculated as last trade price times shares traded on day τ , while the summation is taken over all n trading days during the examination period.
- Co-Skewness (Harvey & Siddique, 2000) and Co-Kurtosis (Dittmar, 2002) are the coefficients $\beta_{j,t}^{CS}$ and $\beta_{j,t}^{CK}$ in the regression $r_{j,\tau} r_{f,\tau} = \alpha_{j,t} + \beta_{j,t}^M (r_{M,\tau} r_{f,\tau}) + \beta_{j,t}^{CS} (r_{M,\tau} r_{f,\tau})^2 + \beta_{j,t}^{CK} (r_{M,\tau} r_{f,\tau})^3 + \epsilon_{j,\tau}$, including the market excess return, the squared market excess return, and the cubed market excess return. The regression is estimated using daily returns over the examination period.
- **Demand for lottery** (Bali et al., 2011, "MAX") is the average of the five highest daily returns during the examination period.
- Downside beta (Ang et al., 2006a) is the coefficient $\beta_{j,t}^D$ in the regression $r_{j,\tau} r_{f,\tau} = \alpha_{j,t} + \beta_{j,t}^D(r_{M,\tau} r_{f,\tau}) + \epsilon_{j,\tau}$, using daily returns over the examination period only when the market return is below the average daily market return over that year.
- Idiosyncratic volatility (Ang et al., 2006b, "idio. volatility") is the standard deviation of the residuals $\epsilon_{j,\tau}$ in the Fama & French (1993) 3-factor model $r_{j,\tau} - r_{f,\tau} =$

 $\alpha_{j,t} + \beta_{j,t}^M(r_{M,\tau} - r_{f,\tau}) + \beta_{j,t}^S SMB_{\tau} + \beta_{j,t}^H HML_{\tau} + \epsilon_{j,\tau}$, using daily returns over the examination period. SMB_{τ} and HML_{τ} denote the returns on the Fama & French (1993) factors.

- **Kurtosis** is the stock's scaled fourth moment, computed using daily returns over the examination period.
- Pastor-Stambough liquidity (Pastor & Stambaugh, 2003, "PS liquidity") is the coefficient $\beta_{j,t}^L$ in the following regression $r_{j,\tau} r_{f,\tau} = \alpha_{j,t} + \beta_{j,t}^M (r_{M,\tau} r_{f,\tau}) + \beta_{j,t}^S SMB_{\tau} + \beta_{j,t}^H HML_{\tau} + \beta_{j,t}^L L_{\tau} + \epsilon_{j,\tau}$, where L_{τ} is the liquidity factor provided by Lubos Pastor and $r_{M,\tau} r_{f,\tau} = MKT_{\tau}$, SMB_{τ} , and HML_{τ} are the Fama-French factors provided by Kenneth R. French. We run the regression using the monthly returns during the examination period.
- Skewness (Xu, 2007) is the stock's scaled third moment computed using daily returns over the examination period.
- **Turnover** (Datar et al., 1998) is the number of shares traded in one month divided by the total shares outstanding, averaged over all months in the examination period.
- Volatility (Zhang, 2006) is the stock's standard deviation computed using daily returns over the examination period.
- Volume (Gervais et al., 2001) is the stock's average daily dollar trading volume over the examination period. In regressions, we take the natural logarithm to remove the extreme skewness in this variable.

References

- Adrian, T., & Rosenberg, J. (2008). Stock returns and volatility: Pricing the short-run and long-run components of market risk. *Journal of Finance*, 63(6), 2997–3030.
- Amihud, Y. (2002). Illiquidity and stock returns: Cross-section and time-series effects. Journal of Financial Markets, 5(1), 31–56.
- Ang, A., Chen, J., & Xing, Y. (2006a). Downside risk. Review of Financial Studies, 19(4), 1191–1239.
- Ang, A., Hodrick, R. J., Xing, Y., & Zhang, X. (2006b). The cross-section of volatility and expected returns. *Journal of Finance*, 61(1), 259–299.
- Ang, A., Liu, J., & Schwarz, K. (2010). Using stocks or portfolios in tests of factor models. Working Paper, Columbia University and University of Pennsylvania.
- Bakshi, G., & Kapadia, N. (2003). Delta-hedged gains and the negative market volatility risk premium. *Review of Financial Studies*, 16(2), 527–566.
- Bakshi, G., Kapadia, N., & Madan, D. (2003). Stock return characteristics, skew laws, and the differential pricing of individual equity options. *Review of Financial Studies*, 16(1), 101–143.
- Bali, T., & Zhou, H. (2015). Risk, uncertainty, and expected returns. Journal of Financial and Quantitative Analysis, forthcoming.
- Bali, T. G., Cakici, N., & Whitelaw, R. F. (2011). Maxing out: Stocks as lotteries and the cross-section of expected returns. *Journal of Financial Economics*, 99(2), 427–446.
- Baltussen, G., Van Bekkum, S., & Van Der Grient, B. (2015). Unknown unknowns: Uncertainty about risk and stock returns. *Journal of Financial and Quantitative Analysis, forthcoming.*
- Bansal, R., & Yaron, A. (2004). Risks for the long run: A potential resolution of asset pricing puzzles. *Journal of Finance*, 59(4), 1481–1509.
- Banz, R. W. (1981). The relationship between return and market value of common stocks. *Journal of Financial Economics*, 9(1), 3–18.
- Barndorff-Nielsen, O. E., Hansen, P. R., Lunde, A., & Shephard, N. (2009). Realized kernels in practice: Trades and quotes. *Econometrics Journal*, 12(3), 1–32.
- Barndorff-Nielsen, O. E., & Shephard, N. (2004). Power and bipower variation with stochastic volatility and jumps. Journal of Financial Econometrics, 2(1), 1–37.
- Barndorff-Nielsen, O. E., & Veraart, A. E. (2012). Stochastic volatility of volatility and variance risk premia. *Journal of Financial Econometrics*, 11(1), 1–46.

- Bhandari, L. C. (1988). Debt/equity ratio and expected common stock returns: Empirical evidence. *Journal of Finance*, 43(2), 507–528.
- Black, F., Jensen, M., & Scholes, M. (1972). The capital asset pricing model: Some empirical tests. In M. Jensen (Ed.) Studies in the Theory of Capital Markets, (pp. 79–121). New York: Praeger.
- Black, F., & Scholes, M. (1973). The pricing of options and corporate liabilities. Journal of Political Economy, 81(3), 637–654.
- Bollerslev, T., Tauchen, G., & Zhou, H. (2009). Expected stock returns and variance risk premia. *Review of Financial Studies*, 22(11), 4463–4492.
- Campbell, J. Y., Giglio, S., Polk, C., & Turley, R. (2014). An intertemporal CAPM with stochastic volatility. *Harvard University Working Paper*.
- Campbell, J. Y., & Shiller, R. J. (1988). The dividend–price ratio and expectations of future dividends and discount factors. *Review of Financial Studies*, 1(3), 195–228.
- Carhart, M. M. (1997). On persistence in mutual fund performance. *Journal of Finance*, 52(1), 57–82.
- Chang, B. Y., Christoffersen, P., & Jacobs, K. (2013). Market skewness risk and the cross section of stock returns. *Journal of Financial Economics*, 107(1), 46–68.
- Chen, N.-F., Roll, R., & Ross, S. A. (1986). Economic forces and the stock market. *Journal* of Business, 59(3), 383–403.
- Chen, T.-F., Chung, S.-L., & Lin, J.-C. (2014). Volatility-of-volatility risk and asset prices. LSU Working Paper.
- Chordia, T., Subrahmanyam, A., & Tong, Q. (2014). Have capital market anomalies attenuated in the recent era of high liquidity and trading activity? *Journal of Accounting and Economics*, 58(1), 41–58.
- Cremers, K., Halling, M., & Weinbaum, D. (2015). Aggregate jump and volatility risk in the cross-section of stock returns. *Journal of Finance*, 70(2), 577–614.
- Datar, V. T., Y Naik, N., & Radcliffe, R. (1998). Liquidity and stock returns: An alternative test. Journal of Financial Markets, 1(2), 203–219.
- D'Avolio, G. (2002). The market for borrowing stock. *Journal of Financial Economics*, 66(2), 271–306.
- Dittmar, R. F. (2002). Nonlinear pricing kernels, kurtosis preference, and evidence from the cross section of equity returns. *Journal of Finance*, (pp. 369–403).
- Epstein, L. G., & Ji, S. (2013). Ambiguous volatility and asset pricing in continuous time. *Review of Financial Studies*, 26(7), 1740–1786.

- Epstein, L. G., & Zin, S. E. (1989). Substitution, risk aversion, and the temporal behavior of consumption and asset returns: A theoretical framework. *Econometrica*, 57(4), 937–969.
- Epstein, L. G., & Zin, S. E. (1991). Substitution, risk aversion, and the temporal behavior of consumption and asset returns: An empirical analysis. *Journal of Political Economy*, 99(2), 263–286.
- Fabozzi, F. J., Huang, D., & Wang, J. (2016). What difference do new factor models make in portfolio allocation? *EDHEC Working Paper*.
- Fama, E. F., & French, K. R. (1992). The cross-section of expected stock returns. Journal of Finance, 47(2), 427–465.
- Fama, E. F., & French, K. R. (1993). Common risk factors in the returns on stocks and bonds. Journal of Financial Economics, 33(1), 3–56.
- Fama, E. F., & French, K. R. (2008). Dissecting anomalies. Journal of Finance, 63(4), 1653–1678.
- Fama, E. F., & French, K. R. (2015). A five-factor asset pricing model. Journal of Financial Economics, 116(1), 1–22.
- Fama, E. F., & MacBeth, J. D. (1973). Risk, return, and equilibrium: Empirical tests. Journal of Political Economy, 81(4), 607–636.
- Frazzini, A., & Pedersen, L. H. (2014). Betting against beta. Journal of Financial Economics, 111(1), 1–25.
- Friend, I., & Blume, M. E. (1975). The demand for risky assets. American Economic Review, 65, 900–922.
- Gervais, S., Kaniel, R., & Mingelgrin, D. H. (2001). The high-volume return premium. Journal of Finance, 56(3), 877–919.
- Han, B., & Zhou, Y. (2012). Variance risk premium and cross-section of stock returns. University of Texas at Austin Working Paper.
- Harvey, C. R., & Siddique, A. (2000). Conditional skewness in asset pricing tests. Journal of Finance, 55(3), 1263–1295.
- Hou, K., Xue, C., & Zhang, L. (2015a). A comparison of new factor models. *Ohio State University Working Paper*.
- Hou, K., Xue, C., & Zhang, L. (2015b). Digesting anomalies: An investment approach. *Review of Financial Studies*, 28(3), 650–705.
- Huang, D., & Shaliastovich, I. (2014). Volatility of volatility risk. University of Pennsylvania Working Paper.

- Jegadeesh, N. (1990). Evidence of predictable behavior of security returns. *Journal of Finance*, 45(3), 881–898.
- Jegadeesh, N., & Titman, S. (1993). Returns to buying winners and selling losers: Implications for stock market efficiency. *Journal of Finance*, 48(1), 65–91.
- Jiang, G. J., & Tian, Y. S. (2005). The model-free implied volatility and its information content. Review of Financial Studies, 18(4), 1305–1342.
- Lettau, M., Maggiori, M., & Weber, M. (2014). Conditional risk premia in currency markets and other asset classes. *Journal of Financial Economics*, 114(2), 197–225.
- Lewellen, J., Nagel, S., & Shanken, J. (2010). A skeptical appraisal of asset pricing tests. Journal of Financial Economics, 96(2), 175–194.
- Lo, A. W., & MacKinlay, A. C. (1990). Data-snooping biases in tests of financial asset pricing models. *Review of Financial Studies*, 3(3), 431–467.
- Merton, R. C. (1980). On estimating the expected return on the market: An exploratory investigation. *Journal of Financial Economics*, 8(4), 323–361.
- Newey, W. K., & West, K. D. (1987). A simple, positive semi-definite, heteroskedasticity and autocorrelation consistent covariance matrix. *Econometrica*, 55(3), 703–708.
- Novy-Marx, R. (2013). The other side of value: The gross profitability premium. *Journal of Financial Economics*, 108(1), 1–28.
- Pastor, L., & Stambaugh, R. F. (2003). Liquidity risk and expected stock returns. Journal of Political Economy, 111(3), 642–685.
- Prokopczuk, M., & Wese Simen, C. (2014). The importance of the volatility risk premium for volatility forecasting. *Journal of Banking & Finance*, 40, 303–320.
- Shanken, J. (1992). On the estimation of beta-pricing models. Review of Financial Studies, 5(1), 1–55.
- Shumway, T. (1997). The delisting bias in CRSP data. Journal of Finance, 52(1), 327–340.
- Shumway, T., & Warther, V. A. (1999). The delisting bias in CRSP's NASDAQ data and its implications for the size effect. *Journal of Finance*, 54(6), 2361–2379.
- Sims, C. A. (1980). Macroeconomics and reality. *Econometrica*, 48, 1–48.
- Xu, J. (2007). Price convexity and skewness. Journal of Finance, 62(5), 2521–2552.
- Zhang, X. (2006). Information uncertainty and stock returns. *Journal of Finance*, 61(1), 105–137.



Figure 1: Time Series of VIX and VVIX



Figure 2: Time Series of Cumulative Hedge Portfolio Returns

itelun

Table 1: Sample Correlations of Different Aggregate Factors	is table presents the sample correlation coefficients of the aggregate factors dVVIX, dVIX, straddle vol, dSkew, dKurt, jump, dVRP,	$\mathrm{KT},\mathrm{SMB},\mathrm{HML},\mathrm{momentum},\mathrm{SMB}^{HXZ},\mathrm{IA},\mathrm{ROE},\mathrm{dVVIX}_{\mathrm{ARMA}},\mathrm{and}\mathrm{dVoVIX}.$ Detailed variable definitions are provided in the appendix.
	Th	MI

	dVVIX	dVIX	Straddle vol	dSkew	dKurt	Jump	dVRP	MKT	SMB	HML	Momentum	SMB^{HXZ}	IA	ROE	$dVVIX_{ARMA}$	dVoVIX
XIVoVb	0.10	0.07	-0.11	-0.05	0.02	0.19	0.01	-0.07	0.03	-0.01	-0.02	0.01	-0.02	0.00	0.11	*
аиулу _{АМЯА} ХIVVb	1.00	0.66	-0.06	0.08	-0.16	0.46	-0.37	-0.55	-0.09	-0.12	0.02	-0.14	0.10	0.11	*	
ROE	0.10	0.18	-0.03	0.06	-0.07	0.07	-0.12	-0.18	-0.08	-0.06	0.15	-0.13	0.20	*		
ΥI	0.10	0.21	-0.01	0.08	-0.09	0.06	-0.18	-0.23	-0.05	0.19	0.12	0.04	*			
$_{ZXH}$ BWS	-0.13	-0.18	-0.05	-0.07	0.11	-0.07	0.06	0.33	0.78	0.36	-0.33	*				
mutnəmoM	0.02	0.22	0.05	0.12	-0.09	-0.02	-0.22	-0.39	0.03	-0.59	*					
TWH	-0.12	-0.27	-0.03	-0.09	0.06	-0.06	0.22	0.40	-0.10	*						
SMB	-0.09	-0.07	-0.03	-0.01	0.07	-0.05	-0.02	0.17	*							
TXM	-0.55	-0.84	-0.02	-0.24	0.30	-0.27	0.64	*								
ЧЛВР	-0.37	-0.77	-0.01	-0.11	0.14	-0.20	*									
dumf	0.46	0.43	-0.69	0.05	-0.09	*										
тшЯb	-0.16	-0.22	-0.04	-0.89	*											
мәҳЅҌ	0.09	0.15	0.03	*												
lov slbbært2	-0.06	-0.05	*													
ХІЛР	0.67	*														
ХІЛЛР	*															

Table 2: Summary Statistics

Panel A of this table presents summary statistics on the aggregate factors dVVIX, dVIX, straddle vol, dSkew, dKurt, jump, dVRP, MKT, SMB, HML, momentum, SMB^{HXZ}, IA, ROE, dVVIX_{ARMA}, and dVoVIX. Detailed variable definitions are provided in the appendix. Panel B provides yearly summary statistics on the VVIX and Panel C shows yearly summary statistics on the individual stocks' sensitivities to aggregate volatility-of-volatility, β^{Q} , with the sensitivity estimation starting in the year denoted first in the first column. Mean, Median, and Std. dev. refer to the sample average, median, and standard deviation of the factors, respectively. P10 and P90 refer to the 10% and 90% percentiles, respectively. AR(1) indicates the first order autocorrelation.

Panel A: Market Factors										
Variable	Mean	Median	Std. dev.	AR(1)	P10	P90				
dVVIX	0.00011	-0.00350	0.0471	-0.0838	-0.0471	0.0511				
dVIX	0.00001	-0.00095	0.0192	-0.1284	-0.0170	0.0175				
Straddle vol	-0.00014	0.00109	0.0158	-0.0408	-0.0166	0.0156				
dSkew	-0.00228	-0.00351	0.0790	0.2265	-0.0969	0.0896				
dKurt	0.00109	-0.00232	0.1205	0.1339	-0.1369	0.1454				
Jump	-0.00171	-0.00962	0.0557	-0.0953	-0.0447	0.0428				
dVRP	0.00000	-0.00002	0.0086	-0.0379	-0.0039	0.0037				
MKT	0.00033	0.00080	0.0129	-0.0841	-0.0130	0.0128				
SMB	0.00004	0.00010	0.0059	-0.0430	-0.0069	0.0067				
HML	-0.00003	-0.00020	0.0069	-0.0251	-0.0063	0.0061				
Momentum	-0.00001	0.00050	0.0104	0.1214	-0.0102	0.0099				
SMB^{HXZ}	0.00013	0.00004	0.0062	0.0057	-0.0072	0.0068				
IA	0.00017	0.00008	0.0042	0.0541	-0.0043	0.0049				
ROE	0.00040	0.00038	0.0044	0.0191	-0.0043	0.0054				
dVVIX _{ARMA}	0.00127	-0.00226	0.0498	0.0005	-0.0481	0.0540				
dVoVIX	0.00022	0.00503	0.0421	0.1784	-0.0338	0.0295				

Year	Mean	Median	Std. dev.	P10	P90						
2006	0.8024	0.8073	0.1126	0.6601	0.9467						
2007	0.8768	0.8618	0.1331	0.7194	1.0469						
2008	0.8185	0.7741	0.1560	0.6763	1.1088						
2009	0.7978	0.7913	0.0863	0.6954	0.9225						
2010	0.8836	0.8622	0.1307	0.7538	1.0346						
2011	0.9294	0.9146	0.1021	0.8182	1.0491						
2012	0.9484	0.9384	0.0838	0.8416	1.0740						
2013	0.8052	0.7960	0.0897	0.6967	0.9203						
2014	0.8301	0.7991	0.1433	0.6771	0.9990						
2015	0.9482	0.9041	0.1475	0.8002	1.1533						
2016	0.9258	0.8981	0.1021	0.8260	1.0949						
total	0.8696	0.8523	0.1332	0.7100	1.0450						

Table 2: Summary Statistics (continued)

Panel B: VVIX Summary Statistics

Panel C: $\beta^{\mathbf{Q}}$ Summary Statistics

Year	Mean	Median	Std. dev.	P10	P90					
2006-2007	0.0093	0.0074	0.0502	-0.0425	0.0650					
2007-2008	0.0085	0.0072	0.0730	-0.0693	0.0904					
2008-2009	0.0433	0.0367	0.1050	-0.0619	0.1585					
2009-2010	0.0066	0.0053	0.0687	-0.0593	0.0751					
2010-2011	0.0105	0.0093	0.0521	-0.0443	0.0675					
2011-2012	0.0131	0.0118	0.0559	-0.0455	0.0766					
2012-2013	0.0037	0.0018	0.0627	-0.0617	0.0729					
2013-2014	-0.0054	-0.0040	0.0648	-0.0724	0.0581					
2014 - 2015	-0.0057	-0.0043	0.0590	-0.0631	0.0489					
2015-2016	-0.0060	-0.0073	0.0626	-0.0654	0.0559					
total	0.0078	0.0049	0.0684	-0.0589	0.0789					

Table 3: Predictability of the VIX and VVIX

This table presents the results of monthly non-overlapping predictive regressions for the realized variance of the S&P 500 and the VIX over the following month. Constant refers to the regression intercept while Slope VIX and Slope VVIX denote the regression slope coefficients on the VIX and VVIX, respectively. adj. R^2 denotes the adjusted R^2 of the regression. Robust Newey & West (1987) standard errors (s.e.) using 4 lags are reported in parentheses. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

	Vol	atility Foreca	sting	Volatility-of-Volatility Forecasting			
Constant (s.e.) Slope VIX (s.e.) Slope VVIX (s.e.)	-0.0190 (0.040) 0.8847*** (0.047)	-0.0173 (0.076) 0.2068^{**} (0.094)	$\begin{array}{c} -0.0185 \\ (0.040) \\ 0.8850^{***} \\ (0.124) \\ -0.0007 \\ (0.050) \end{array}$	$\begin{array}{c} 0.3035^{***}\\ (0.070)\\ -0.0139\\ (0.084) \end{array}$	$\begin{array}{c} 0.0215 \\ (0.046) \end{array}$ $\begin{array}{c} 0.3238^{***} \\ (0.054) \end{array}$	$\begin{array}{c} 0.0217 \\ (0.046) \\ -0.1753^{*} \\ (0.096) \\ 0.3649^{***} \\ (0.059) \end{array}$	
adj. \mathbb{R}^2	0.583	0.053	0.580	-0.008	0.135	0.148	

Table 4: Portfolios Sorted by Exposure to Aggregate Volatility-of-Volatility

At the beginning of each month, we form quintile portfolios based on the stock's sensitivities to innovations in aggregate volatility-of-volatility $(\beta_{j,t}^{\mathbf{Q}})$ over the following year. To obtain the sensitivities, we regress daily excess stock returns on dVVIX, controlling for MKT and dVIX as in Equation (14). Stocks with the lowest $\beta_{j,t}^{\mathbf{Q}}$ are sorted into portfolio 1, those with the highest $\beta_{j,t}^{\mathbf{Q}}$ into portfolio 5. The column labeled 5 minus 1 refers to the hedge portfolio buying the quintile of stocks with the highest $\beta_{j,t}^{\mathbf{Q}}$ and simultaneously selling the stocks in the quintile with the lowest $\beta_{j,t}^{\mathbf{Q}}$ Each month, we set up new 12-month portfolios. The row labeled Mean return is based on monthly simple returns. CAPM alpha, 4-factor alpha, and 5-factor alpha refer to the alphas of the CAPM. the Carhart (1997) 4-factor, and the 5-factor models (including Pastor & Stambaugh (2003) liquidity), respectively. HXZ alpha denotes the alpha relative to the Hou et al. (2015b) factor model. The segment Factor loadings denotes the value-weighted average annual factor loadings, where β^{M} $\beta^{\rm V}$, and $\beta^{\rm Q}$ refer to the factor loadings on the market factor, dVIX, and dVVIX, respectively. The segment Stock characteristics presents average (value-weighted) portfolio characteristics with Mkt. share denoting the average market share of the portfolios. The remaining variable definitions are provided in the Appendix. Robust Newey & West (1987) p-values using 12 lags are reported in parentheses. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

Rank	1	2	3	4	5	5 minus 1
equally weighted						
Mean return	0.1004	0.1062**	0.0987^{*}	0.0826	0.0514	-0.0491**
	(0.156)	(0.049)	(0.069)	(0.141)	(0.460)	(0.027)
CAPM alpha	-0.0106	0.0201	0.0123	-0.0064	-0.0593**	-0.0487**
	(0.699)	(0.113)	(0.292)	(0.629)	(0.014)	(0.021)
4-factor alpha	-0.0072	0.0189***	0.0147***	-0.0052*	-0.0563***	-0.0491***
	(0.504)	(0.000)	(0.000)	(0.081)	(0.000)	(0.004)
5-factor alpha	-0.0023	0.0194***	0.0137***	-0.0049	-0.0485***	-0.0462**
Ĩ	(0.840)	(0.000)	(0.000)	(0.120)	(0.000)	(0.022)
HXZ alpha	0.0103	0.0264***	0.0153***	-0.0074	-0.0532***	-0.0635**
-	(0.457)	(0.000)	(0.004)	(0.305)	(0.000)	(0.015)
value-weighted						
Mean return	0.0967	0.0951**	0.0914**	0.0652	0.0433	-0.0533*
	(0.111)	(0.047)	(0.044)	(0.198)	(0.518)	(0.089)
CAPM alpha	-0.0003	0.0193***	0.0185***	-0.0140	-0.0629***	-0.0626*
	(0.985)	(0.000)	(0.000)	(0.121)	(0.007)	(0.056)
4-factor alpha	-0.0039	0.0161^{***}	0.0195^{***}	-0.0091	-0.0507***	-0.0468*
	(0.755)	(0.001)	(0.000)	(0.196)	(0.004)	(0.075)
5-factor alpha	0.0063	0.0194^{***}	0.0144^{***}	-0.0126*	-0.0508**	-0.0572^{*}
	(0.553)	(0.000)	(0.000)	(0.058)	(0.015)	(0.057)
HXZ alpha	0.0198	0.0204^{***}	0.0173^{***}	-0.0148	-0.0657^{***}	-0.0855**
	(0.199)	(0.000)	(0.000)	(0.264)	(0.009)	(0.029)
Factor loadings						
β^{M}	1.0708***	0.9253***	0.9729***	1.0989^{***}	1.3848^{***}	0.3140***
1	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
β^{V}	0.1441***	0.0395^{***}	-0.0041	-0.0401***	-0.0907***	-0.2349***
	(0.000)	(0.000)	(0.150)	(0.000)	(0.000)	(0.000)
eta^{Q}	-0.0599***	-0.0177***	0.0062*	0.0313***	0.0777***	0.1376***
	(0.000)	(0.000)	(0.057)	(0.000)	(0.000)	(0.000)
Return characteristics						
Std. deviation	0.0601	0.0474	0.0450	0.0504	0.0668	0.0311
Skewness	-0.1632	-0.4981	-0.8644	-0.8132	-0.0727	0.0175
Kurtosis	4.0501	3.6484	3.9630	3.6487	3.8287	2.3774
Mkt. share	0.2042	0.2467	0.2209	0.2034	0.1248	-0.0794
$\log(\text{Size})$	16.863	17.444	17.330	17.087	16.314	-0.5490
Book-to-market	0.5416	0.4850	0.4805	0.5196	0.6716	0.1299
Bid-ask spread	0.0008	0.0006	0.0006	0.0006	0.0009	0.0002
Age	33.491	41.099	38.442	35.074	28.564	-4.9267
Leverage	0.5567	0.5549	0.5622	0.5861	0.6015	0.0448

Table 4: Portfolios Sorted by Exposure to Aggregate Volatility-of-Volatility (continued)

Table 5: Dissecting the Effects of Aggregate Risk and Aggregate Volatility-of-Volatility

This table reports Hou et al. (2015b) factor alphas for double-sorted portfolios. We obtain the factor sensitivities in a joint multivariate estimation. At the beginning of each month, we independently sort stocks into quintiles based on their sensitivity to aggregate volatility (Panels A and B) and the market variance risk premium (Panels C and D) as well as their volatility-of-volatility-sensitivities $(\beta_{i,t}^{\mathbf{Q}})$ using NYSE breakpoints. We report the alphas of all 25 portfolios as well as the respective long minus short (5 minus 1) portfolios. Each month, we set up new 12-month portfolios. Robust Newey & West (1987) p-values using 12 lags are reported in parentheses. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

	Panel A: Aggregate Volatility – Equally Weighted										
				_							
		1	2	3	4	5	5 minus 1				
	1	0.0741^{*}	0.0716^{***}	0.0578^{***}	0.0277**	-0.0497***	-0.1238***				
		(0.053)	(0.000)	(0.000)	(0.036)	(0.000)	(0.000)				
	2	0.0777^{***}	0.0684^{***}	0.0591^{***}	0.0068	-0.0566***	-0.1343***				
		(0.000)	(0.000)	(0.000)	(0.268)	(0.000)	(0.000)				
AVVIX	3	0.0653***	0.0443***	0.0288***	0.0041	-0.0699***	-0.1352***				
UVVIA		(0.000)	(0.000)	(0.000)	(0.618)	(0.000)	(0.000)				
	4	0.0454^{***}	0.0327***	0.0111	-0.0307**	-0.0920***	-0.1374***				
		(0.000)	(0.000)	(0.165)	(0.016)	(0.000)	(0.000)				
	5	-0.0075	-0.0339***	-0.0523***	-0.0810***	-0.1277***	-0.1202***				
		(0.510)	(0.001)	(0.000)	(0.001)	(0.001)	(0.004)				
	5 minus 1	-0.0817*	-0.1055***	-0.1101***	-0.1086***	-0.0781*					
		(0.058)	(0.000)	(0.000)	(0.001)	(0.060)					

Panel B: Aggregate Volatility – Value-Weighted

				dVIX			_
		1	2	3	4	5	5 minus 1
	1	0.1007***	0.0664^{***}	0.0394***	0.0135	-0.0803***	-0.1810***
		(0.001)	(0.000)	(0.006)	(0.335)	(0.000)	(0.000)
	2	0.0794^{***}	0.0574^{***}	0.0278***	-0.0250***	-0.0547***	-0.1341***
		(0.000)	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)
AVVIV	3	0.0594^{***}	0.0315^{***}	0.0241^{***}	-0.0118	-0.0978***	-0.1573***
		(0.000)	(0.000)	(0.000)	(0.192)	(0.000)	(0.000)
	4	0.0360^{***}	0.0032	-0.0017	-0.0507**	-0.1186***	-0.1546***
		(0.007)	(0.737)	(0.912)	(0.016)	(0.000)	(0.000)
	5	-0.0217	-0.0305	-0.0769***	-0.0915***	-0.1444***	-0.1227***
		(0.128)	(0.220)	(0.000)	(0.005)	(0.000)	(0.002)
	5 minus 1	-0.1224***	-0.0969**	-0.1163***	-0.1051**	-0.0641	
		(0.003)	(0.018)	(0.000)	(0.012)	(0.145)	

					1 0	0				
			dVRP							
		1	2	3	4	5	5 minus 1			
	1	-0.0270	0.0312***	0.0505***	0.0703***	0.0993***	0.1262***			
	2	(0.128) -0.0361*** (0.000)	(0.009) 0.0181^{***} (0.003)	(0.000) 0.0420^{***} (0.000)	(0.000) 0.0665^{***} (0.000)	(0.000) 0.0573^{***} (0.000)	(0.000) 0.0934^{***} (0.000)			
AWWIX	3	-0.0301***	0.0024	0.0314***	0.0356***	0.0530***	0.0831***			
UVVIX	4	(0.003) - 0.0606^{***} (0.000)	(0.739) -0.0313*** (0.004)	(0.000) 0.0014 (0.871)	(0.000) 0.0201^{***} (0.004)	(0.000) 0.0281^{***} (0.000)	(0.000) 0.0886^{***} (0.000)			
	5	-0.1294*** (0.000)	-0.0721*** (0.000)	-0.0568*** (0.004)	-0.0473*** (0.000)	-0.0410*** (0.001)	0.0884^{***} (0.004)			
	5 minus 1	-0.1025^{***} (0.007)	-0.1033^{***} (0.000)	-0.1073^{***} (0.000)	-0.1176^{***} (0.000)	-0.1403^{***} (0.000)				

Table 5: Dissecting the Effects of Aggregate Risk and Aggregate Volatility-of-Volatility (continued)

Panel C	: Market	Variance	Risk	Premium -	Equally	Weighted
---------	----------	----------	------	-----------	---------	----------

Panel D: Market Variance Risk Premium – Value-Weighted

		1	2	3	4	5	5 minus 1
	1	-0.0770***	0.0047	0.0455***	0.0641***	0.1059***	0.1829***
	2	(0.000) - 0.0603^{***} (0.000)	(0.766) -0.0133 (0.113)	(0.000) 0.0146^{**} (0.016)	(0.000) 0.0440^{***} (0.000)	(0.000) 0.0822^{***} (0.000)	(0.000) 0.1424^{***} (0.000)
dVVIX	3	-0.0785^{***}	-0.0050 (0.673)	(0.0237^{**})	0.0220^{***}	0.0461^{***}	0.1245^{***}
	4	-0.0890***	-0.0446^{**}	(0.012) -0.0089 (0.601)	(0.002) -0.0071 (0.587)	0.0298***	0.1188***
	5	(0.000) -0.1386^{***} (0.000)	(0.031) -0.0977^{***} (0.000)	(0.001) -0.0731** (0.014)	(0.387) -0.0604^{**} (0.010)	(0.003) -0.0799^{***} (0.000)	(0.000) 0.0587^{*} (0.084)
	5 minus 1	-0.0616^{*} (0.081)	-0.1025^{***} (0.006)	-0.1186^{***} (0.002)	-0.1245^{***} (0.000)	-0.1859^{***} (0.000)	

Regressions	
MacBeth	
Fama-l	
. 0	
Table	

using 12 lags, that also incorporate the Shanken (1992) errors-in-variables correction, are reported in parentheses. *, **, and *** indicate This table presents average coefficient estimates from monthly Fama & MacBeth (1973) regressions. Each month, we regress excess stock returns during the following year on a constant, the market betas, the sensitivities to innovations in aggregate volatility, and the sensitivities to innovations in aggregate volatility-of-volatility over the same time, as well as a series of stock characteristics, all also measured over the following year. Detailed variable definitions are provided in the appendix. Robust Newey & West (1987) p-values significance at 10%, 5%, and 1%, respectively.

Factor	(i)	(ii)	(iii)	(iv)	(v)	(vi)	(vii)	(viii)	(ix)
Constant	0.1005^{***}	-0.4661^{***}	0.1202^{***}	0.1226^{***}	0.1077^{***}	0.1009^{***}	0.1002^{***}	0.0807^{**}	-0.3124^{***}
	(0.004)	(0.000)	(0.001)	(0.00)	(0.002)	(0.003)	(0.002)	(0.020)	(0.000)
Beta	0.0000	0.0191	0.0018	0.0030	-0.0206	-0.0048	-0.0010	0.0015	-0.0008
	(0.999)	(0.767)	(0.977)	(0.962)	(0.723)	(0.939)	(0.987)	(0.981)	(0.989)
dVIX	-0.2279^{**}	-0.2015^{**}	-0.2243**	-0.2232**	-0.2138^{**}	-0.2251^{**}	-0.2289^{**}	-0.2305^{**}	-0.1851^{**}
	(0.017)	(0.037)	(0.020)	(0.019)	(0.013)	(0.016)	(0.015)	(0.016)	(0.033)
dVVIX	-0.7753^{**}	-0.6954^{**}	-0.7532**	-0.7680**	-0.7710^{***}	-0.7736^{***}	-0.8065***	-0.7680^{**}	-0.6886**
	(0.010)	(0.025)	(0.015)	(0.012)	(0.008)	(0.00)	(0.008)	(0.012)	(0.019)
$\ln(Size)$		0.0375^{***}							0.0304^{***}
		(0.00)							(0.00)
Book-to-market		r.	-0.0363^{***}						-0.0350^{***}
			(0.00)						(0.00)
Bid-ask spread				-18.085^{***}					-4.3590^{***}
				(0.000)					(0.004)
Momentum					-0.0510				-0.0698
					(0.304)				(0.157)
Short-term reversal						-0.0496			-0.0903
						(0.324)			(0.115)
$\ln(Age)$							0.0008		-0.0117^{**}
							(0.879)		(0.040)
Leverage								0.0341	0.0164
								(0.172)	(0.559)
adj. \mathbb{R}^2	0.0928	0.1097	0.1047	0.1019	0.1057	0.0987	0.0982	0.0977	0.1435

0.0406^{*} (0.098)	0.0821^{**} (0.022)	0.1046^{***} (0.002)	0.0739^{*} (0.056)	0.0931^{**} (0.011)	0.0832^{**} (0.015)	0.1020^{***} (0.002)	0.0835^{**} (0.025)	Constant
(viii)	(vii)	(vi)	(v)	(iv)	(iii)	(ii)	(i)	
			pectively.	%, and 1%, res	ance at 10%, 5 ⁽	indicate signific	*, **, and ***	in parentheses.
n, are reported	riables correctio	92) errors-in-vai	ie Shanken (19	o incorporate th	2 lags, that also	p-values using 1:	& West (1987) $_{\rm I}$	Robust Newey
the appendix.	are provided in	ble definitions a	Detailed varia	ollowing year.	ata over the fc	, using daily d	n Equation (15)	regression as in
nt multivariate	otained in a joi	$\beta_{j,t}^{\zeta}$ estimates of	or a series of \downarrow	ume time, and/	lity over the sa	latility-of-volati	in aggregate vo	to innovations
he sensitivities	gate volatility, t	vations in aggree	civities to innov	etas, the sensit	t, the market b	ar on a constan	the following ye	returns during
ss excess stock	month, we regre	gressions. Each	Beth (1973) reg	y Fama & Macl	ss from monthly	fficient estimate	ents average coe	This table pres
	n	te Estimatio	- Multivaria	Regressions -	l-MacBeth F	able 7: Fama	Ĥ	

(vii) (viii)	0.0821^{**} 0.0406^{*}	(0.022) (0.098)	0.0369 0.0918*	(0.541) (0.089)	-0.2674^{***}	(0.001)	.7314*** -0.8947***	(0.005) (0.001)	-0.0190	(0.329)	0.0608^{**}	(0.012)								0.0517^{**}	(0.037)	-0.0674**	0.0109	(0.583)	0.0713***	(TOD'D)
(vi)	0.1046^{***} 0	(0.002)	-0.0051	(0.935) (-0.7323** -0.	(0.012)												0.0500** 0	(0.043)					
(v)	0.0739^{*}	(0.056)	0.0456	(0.461)			-0.7615^{***}	(0.005)	-0.0198	(0.304)	-0.0590^{**}	(0.020)					0.1192^{*}	(0.071)	-0.7976** (0.038)							
(iv)	0.0931^{**}	(0.011)	-0.0010	(0.987)			-0.7801^{**}	(0.011)									0.2078^{***}	(0.001)	-1.0184*** (0.003)							
(iii)	0.0832^{**}	(0.015)	0.0361	(0.540)	-0.2179^{**}	(0.016)	-0.7300^{***}	(0.003)	-0.0174	(0.349)	-0.0596^{**}	(0.014)	0.4023	(0.340)	-0.9010	(0.146)										
(ii)	0.1020^{***}	(0.002)	-0.0025	(0.967)	-0.2077**	(0.020)	-0.7539^{***}	(0.007)					0.6213	(0.167)	-1.3352^{*}	(0.073)										
(i)	0.0835^{**}	(0.025)	0.0354	(0.560)	-0.2147^{**}	(0.022)	-0.7101^{***}	(0.007)	-0.0186	(0.340)	-0.0599**	(0.015)														
	Constant		MKT		dVIX		dVVIX		SMB		HML		dSkew		dKurt		Straddle vol	,	dunf	dVRP	2711	SMB^{HAZ}	IA		ROE	

Factors
onalized
- Orthog
Regressions –
Fama-MacBeth]
Table 8:

This table presents average coefficient estimates from monthly Fama & MacBeth (1973) regressions using orthogonalized factors. Each month, we regress excess stock returns during the following year on a constant, the market betas, the sensitivities to innovations in aggregate volatility, and the sensitivities to innovations in aggregate volatility-of-volatility over the same time, as well as a series of stock characteristics, all also measured over the following year. Detailed variable definitions are provided in the appendix. Robust Newey & West (1987) p-values using 12 lags, that also incorporate the Shanken (1992) errors-in-variables correction, are reported in parentheses. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

Factor	(i)	(ii)	(iii)	(iv)	(v)	(vi)	(vii)	(viii)	(ix)
Constant	0.1038^{***}	-0.4782***	0.1230^{***}	0.1263^{***}	0.1103^{***}	0.1041^{***}	0.1016^{***}	0.0834^{**}	-0.3288***
	(0.003)	(0.000)	(0.001)	(0.00)	(0.002)	(0.002)	(0.002)	(0.019)	(0.000)
Beta	-0.0073	0.0124	-0.0046	-0.0044	-0.0277	-0.0119	-0.0074	-0.0055	-0.0074
	(0.910)	(0.852)	(0.943)	(0.946)	(0.642)	(0.851)	(0.907)	(0.932)	(0.903)
dVIX	-0.2076^{***}	-0.1550^{***}	-0.2037***	-0.1983^{***}	-0.2168^{***}	-0.2093***	-0.2098***	-0.2086***	-0.1605^{***}
	(0.00)	(0.000)	(0.000)	(0.00)	(0.000)	(0.000)	(0.00)	(0.000)	(0.00)
dVVIX	-0.3829^{*}	-0.3509^{*}	-0.3688^{*}	-0.3872^{*}	-0.3958^{**}	-0.3869*	-0.4086^{**}	-0.3732^{*}	-0.3688^{*}
	(0.056)	(0.088)	(0.070)	(0.055)	(0.042)	(0.051)	(0.037)	(0.068)	(0.060)
$\ln(Size)$		0.0385^{***}							0.0317^{***}
		(0.000)							(0.00)
Book-to-market		~	-0.0365^{***}						-0.0348^{***}
			(0.00)						(0.00)
Bid-ask spread				-18.216^{***}					-4.3741^{***}
				(0.00)					(0.002)
Momentum					-0.0501				-0.0695
					(0.312)				(0.160)
Short-term reversal						-0.0478			-0.0891
						(0.334)			(0.117)
$\ln({ m Age})$							0.0012		-0.0118^{**}
							(0.817)		(0.043)
Leverage								0.0349	0.0166
								(0.150)	(0.547)
adj. \mathbb{R}^2	0.0948	0.1125	0.1064	0.1040	0.1073	0.1006	0.1000	0.0997	0.1453

Table 9: Pre-Formation Factor Loadings

At the beginning of each month, we form quintile portfolios based on the stock's sensitivities to aggregate volatility-of-volatility $(\beta_{j,t}^{\mathbf{Q}})$ over the past year. To obtain the sensitivities, we regress daily excess stock returns on dVVIX, controlling for MKT and dVIX as in Equation (14). Stocks with the lowest $\beta_{j,t}^{\mathbf{Q}}$ are sorted into portfolio 1, those with the highest $\beta_{j,t}^{\mathbf{Q}}$ into portfolio 5. The column labeled 5 minus 1 refers to the hedge portfolio buying the quintile of stocks with the highest $\beta_{j,t}^{\mathbf{Q}}$ and simultaneously selling the stocks in the quintile with the lowest $\beta_{j,t}^{\mathbf{Q}}$. Each month, we set up new 12-month portfolios. We report Hou et al. (2015b) factor alphas for portfolios based on returns over the following year. Robust Newey & West (1987) p-values using 12 lags are reported in parentheses. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively. The columns ex post $\beta^{\mathbf{Q}}$ report the average annual ex post sensitivities to aggregate volatility-of-volatility of the portfolios.

Rank	1	2	3	4	5	5 minus 1
equally weighted						
Mean return	0.0891	0.0918^{*}	0.0871	0.0804	0.0997	0.0106
	(0.123)	(0.071)	(0.113)	(0.174)	(0.147)	(0.682)
HXZ alpha	0.0047	0.0208***	0.0210***	0.0038	0.0106	0.0059
	(0.742)	(0.003)	(0.000)	(0.460)	(0.341)	(0.788)
ex post β^{V}	0.0082	0.0055	0.0063	0.0084	0.0101	0.0019
value-weighted						
Mean return	0.0807	0.0929	0.1019	0.0979	0.0971	0.0164
	(0.242)	(0.137)	(0.105)	(0.125)	(0.171)	(0.253)
HXZ alpha	-0.0084	0.0071	0.0132**	0.0091*	-0.0009	0.0075
	(0.289)	(0.252)	(0.016)	(0.098)	(0.909)	(0.580)
ex post β^{V}	0.0068	0.0025	0.0005	-0.0010	0.0011	-0.0057

Table 10: Determinants of Relative Sensitivities to Aggregate Volatility-of-Volatility

This table presents average coefficient estimates from monthly Fama & MacBeth (1973) regressions. Each month, we regress the sensitivity to innovations in aggregate volatility-of-volatility during the following year on a constant and a series of stock characteristics, all measured over the past year. Detailed variable definitions are provided in the appendix. Robust Newey & West (1987) p-values using 12 lags, that also incorporate the Shanken (1992) errors-in-variables correction, are reported in parentheses. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

Factor	(i)	(ii)	(iii)	(iv)	(v)	(vi)	(vii)
Constant	-0.0068**	0.0136	-0.0062*	-0.0066**	-0.0063**	-0.0072**	0.0288
	(0.023)	(0.509)	(0.068)	(0.026)	(0.044)	(0.038)	(0.287)
Beta	0.0115***	0.0107***	0.0115***	0.0115***	0.0157^{**}	0.0091***	0.0253***
	(0.000)	(0.000)	(0.000)	(0.000)	(0.019)	(0.000)	(0.001)
dVIX	-0.0010	-0.0002	-0.0009	0.0012	-0.0059	0.0046	-0.0164
	(0.779)	(0.943)	(0.840)	(0.762)	(0.465)	(0.265)	(0.119)
dVVIX	0.0100	0.0074	0.0121	0.0142	0.0064	0.0207	-0.0078
	(0.675)	(0.744)	(0.641)	(0.507)	(0.758)	(0.380)	(0.736)
$\ln(\text{Size})$	× ,	-0.0014	× ,	· · · ·		· · · ·	-0.0021
		(0.263)					(0.180)
Book-to-market		0.0000					-0.0009
		(0.997)					(0.408)
Bid-ask spread		· · ·	-0.0523				-0.3474*
-			(0.859)				(0.078)
Momentum			-0.0054***				-0.0064***
			(0.001)				(0.000)
Short-term reversal			-0.0030				-0.0075**
			(0.371)				(0.036)
Idio. volatility				-0.0097			-0.4570*
-				(0.908)			(0.090)
Co-Skewness				0.0000			-0.0001
				(0.730)			(0.569)
Co-Kurtosis				0.0000			0.0000
				(0.671)			(0.419)
Downside beta				· · ·	-0.0054		-0.0170**
					(0.388)		(0.028)
MAX					0.0012		0.0255
					(0.952)		(0.665)
Volatility					. ,	0.1484	0.2827
-						(0.283)	(0.126)
Skewness						0.0005	-0.0002
						(0.564)	(0.763)
Kurtosis						-0.0001*	-0.0001
						(0.094)	(0.149)
adj. R ²	0.0179	0.0285	0.0265	0.0250	0.0228	0.0459	0.0830

Table 11: Delta-Hedged Option Returns

This table presents the delta-hedged option returns of VIX-options. We use options with 30 days to maturity and hold these until expiration. The positions are dynamically delta-hedged using the Black & Scholes (1973) delta. We separate the options into puts and calls as well as different moneyness bins, where moneyness is defined as the strike price K over the futures price F_t at initiation. Mean denotes the average return, computed as the strategy's payoff over the price of the corresponding VIX-future at initiation. T-stat denotes the t-statistic using robust Newey & West (1987) standard errors with 4 lags. Std, SR, Skew, and Kurt denote the standard deviation of returns, its Sharpe ratio, skewness, and kurtosis, respectively. Additionally, Median and the 5% and 95% quantiles, $q^{0.05}$ and $q^{0.95}$, present further details on the distribution of returns. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

	Moneyness $\left(\frac{K}{F_t}\right)$	Mean	T-stat	Std.	SR	Skew	Kurt	Median	$q^{0.05}$	$q^{0.95}$
	[0.7; 0.8]	-0.0008	-0.67	0.013	-0.06	0.74	6.60	-0.0004	-0.0199	0.0225
	[0.8; 0.9]	-0.0020	-1.26	0.021	-0.09	1.15	7.93	-0.0021	-0.0332	0.0325
$C_{\rm all}$	[0.9; 1.0]	-0.0015	-0.59	0.030	-0.05	1.68	8.76	-0.0087	-0.0376	0.0507
Cans	[1.0; 1.1]	-0.0039	-1.28	0.036	-0.11	1.45	5.39	-0.0119	-0.0466	0.0783
	[1.1; 1.2]	-0.0061**	-2.50	0.031	-0.20	1.35	5.81	-0.0119	-0.0451	0.0583
	[1.2; 1.3]	-0.0072**	-2.59	0.029	-0.24	1.59	8.00	-0.0104	-0.0421	0.0615
	[0.7; 0.8]	-0.0027**	-2.20	0.013	-0.20	0.79	7.52	-0.0030	-0.0220	0.0134
	[0.8; 0.9]	-0.0029**	-2.03	0.020	-0.14	0.99	7.31	-0.0034	-0.0346	0.0312
Pute	[0.9; 1.0]	-0.0021	-0.92	0.029	-0.07	1.68	9.13	-0.0081	-0.0388	0.0472
1 uts	[1.0; 1.1]	-0.0051*	-1.76	0.035	-0.14	1.43	5.38	-0.0128	-0.0475	0.0778
	[1.1; 1.2]	-0.0069***	-2.78	0.031	-0.22	1.12	5.36	-0.0121	-0.0468	0.0569
	[1.2; 1.3]	-0.0069**	-2.58	0.029	-0.24	1.47	8.96	-0.0107	-0.0415	0.0523

How Aggregate Volatility-of-Volatility Affects Stock Returns

Online Appendix

JEL classification: G12, G11

Keywords: Aggregate economic uncertainty, stock market volatility-of-volatility, VVIX

$\operatorname{Factors}$	
Aggregate	
Different	
\mathbf{to}	
Sensitivities	
the	
ē	
Correlations	
Rank	
<u>-</u>	
P 0	
Table	

This table presents the average Spearman rank correlation coefficients of the sensitivities of individual stocks to the aggregate factors dVVIX, dVIX, dVIX, dVIX, straddle vol, dSkew, dKurt, jump, dVRP, MKT, SMB, HML, momentum, SMB^{HXZ} , IA, ROE, $dVVIX_{ARMA}$, and dVoVIX. Detailed variable definitions are provided in Appendix A of the main paper.

	dVVIX	dVIX	Straddle vol	dSkew	dKurt	Jump	dVRP	MKT	SMB	HML	Momentum	SMB^{HXZ}	IA	ROE	$dVVIX_{ARMA}$	dVoVIX
ХІЛоДр	0.07	0.07	0.25	-0.03	0.02	-0.17	0.00	0.15	0.14	0.03	-0.06	0.12	-0.01	0.04	0.10	*
_{АМЯА} ХІУУЬ	0.98	-0.40	0.15	-0.01	0.01	-0.04	0.04	0.13	0.11	0.07	-0.07	0.12	0.03	0.04	*	
BOE	0.00	0.07	0.11	-0.01	-0.01	-0.09	-0.05	-0.05	0.12	0.01	0.11	0.10	0.00	*		
ΥI	-0.01	0.00	0.01	0.02	-0.03	-0.02	0.03	-0.14	-0.02	0.32	0.13	-0.01	*			
$_{ZXH}$ BWS	0.10	0.31	0.12	0.02	-0.01	-0.09	-0.31	0.53	0.88	0.24	-0.12	*				
mutnamoM	-0.06	-0.19	-0.18	0.05	-0.07	0.11	0.19	-0.25	-0.04	0.21	*					
IMH	0.05	0.04	0.01	0.03	-0.05	-0.04	-0.04	0.09	0.24	*						
SMB	0.09	0.29	0.12	0.03	-0.02	-0.10	-0.29	0.48	*							
MKT	0.13	0.50	0.36	-0.02	0.00	-0.20	-0.53	*								
ЧЛВР	0.03	-0.74	-0.23	0.07	-0.07	0.05	*									
dunr	-0.05	-0.19	-0.71	0.00	-0.01	*										
truAb	0.01	0.05	-0.02	-0.86	*											
мэяЗb	-0.01	-0.06	0.01	*												
lov slbbsrt2	0.15	0.37	*													
ХІЛР	-0.40	*														
ХІЛЛР	*															

Table A2: Fama–MacBeth Regressions (Value-Weighted)
This table presents average coefficient estimates from monthly value-weighted Fama & MacBeth (1973) regressions. Each month, we regress
excess stock returns during the following year on a constant, the market betas, the sensitivities to innovations in aggregate volatility,
and the sensitivities to innovations in aggregate volatility-of-volatility over the same time, as well as a series of stock characteristics, all
also measured over the following year. Detailed variable definitions are provided in the appendix. Robust Newey & West (1987) p-values
using 12 lags, that also incorporate the Shanken (1992) errors-in-variables correction, are reported in parentheses. *, **, and *** indicate

significance at 10%, 5%, and 1%, respectively.

Factor	(i)	(ii)	(iii)	(iv)	(v)	(vi)	(vii)	(viii)	(ix)
Constant	0.0987***	0.0173	0.1152^{***}	0.1042^{***}	0.1030^{***}	0.0989^{***}	0.1516^{***}	0.1109^{***}	0.0643
	(0.006)	(0.831)	(0.002)	(0.004)	(0.004)	(0.006)	(0.000)	(0.006)	(0.480)
Beta	-0.0035	-0.0020	0.0076	0.001	-0.0201	-0.0070	-0.0117	0.0010	-0.0086
	(0.954)	(0.972)	(0.896)	(0.998)	(0.725)	(0.905)	(0.843)	(0.987)	(0.869)
dVIX	-0.2732^{***}	-0.2686^{***}	-0.2784^{***}	-0.2717***	-0.2628***	-0.2724^{***}	-0.2602^{**}	-0.2791^{***}	-0.2688***
	(0.008)	(0.010)	(0.005)	(0.00)	(0.00)	(0.008)	(0.011)	(0.007)	(0.007)
dVVIX	-0.8377^{**}	-0.8406^{**}	-0.8238**	-0.8336^{**}	-0.8109^{**}	-0.8422^{**}	-0.8016^{**}	-0.8292^{**}	-0.8008**
	(0.017)	(0.020)	(0.016)	(0.018)	(0.024)	(0.016)	(0.026)	(0.018)	(0.024)
$\ln(Size)$		0.0047							0.0071
		(0.320)							(0.170)
Book-to-market			-0.0564^{***}						-0.0611^{***}
			(0.000)						(0.00)
Bid-ask spread				-15.900^{***}					-0.3509
				(0.000)					(0.935)
Momentum					-0.0505				-0.0783*
					(0.256)				(0.098)
Short-term reversal						-0.0621			-0.1007^{*}
						(0.246)			(0.076)
$\ln(Age)$							-0.0135^{***}		-0.0154^{***}
							(0.002)		(0.00)
Leverage								-0.0294	-0.0250
								(0.284)	(0.324)
adj. \mathbb{R}^2	0.1729	0.1804	0.1935	0.1767	0.1924	0.1833	0.1814	0.1822	0.2415

Table A3: Fama–MacBeth Regressions (Further Control Variables)
This table presents average coefficient estimates from monthly Fama & MacBeth (1973) regressions. Each month, we regress excess
stock returns during the following year on a constant, the market betas, the sensitivities to innovations in aggregate volatility, and the
sensitivities to innovations in aggregate volatility-of-volatility over the same time, as well as a series of stock characteristics, all also
measured over the following year. Detailed variable definitions are provided in the Appendix to the main paper. Panel A examines
returns distribution characteristics and Panel B liquidity-related characteristics. Robust Newey & West (1987) p-values using 12 lags,
that also incorporate the Shanken (1992) errors-in-variables correction, are reported in parentheses. *, **, and *** indicate significance
at 10%, 5%, and 1%, respectively.

			<u>Panel A: Returi</u>	ns Distribution	s Characteristic	S		
	(x)	(xi)	(xii)	(xiii)	(xiv)	(xv)	(xvi)	(xvii)
Constant	-0.3112^{**}	-0.3612^{***}	-0.3591^{***}	-0.3541^{***}	0.0696	-0.0953	-0.0716	0.0956
	(0.021)	(0.00)	(0.00)	(0.000)	(0.507)	(0.203)	(0.383)	(0.357)
Beta	0.0288	0.0210	0.0252	0.0341	0.0635	0.0070	0.0057	0.0397
	(0.533)	(0.739)	(0.694)	(0.640)	(0.217)	(0.910)	(0.928)	(0.441)
dVIX	-0.2025^{***}	-0.2055^{**}	-0.1766*	-0.2286^{**}	-0.2548^{***}	-0.1953^{**}	-0.1896^{**}	-0.1377*
	(0.008)	(0.040)	(0.069)	(0.030)	(0.002)	(0.030)	(0.038)	(0.065)
dVVIX	-0.6821^{**}	-0.6947**	-0.6722**	-0.7131^{***}	-0.9411^{***}	-0.7618^{***}	-0.7511^{***}	-0.5932^{***}
	(0.014)	(0.031)	(0.029)	(0.003)	(0.00)	(0.007)	(0.00)	(0.002)
ln(Size)	0.0285^{***}	0.0317^{***}	0.0313^{***}	0.0314^{***}	0.0063	0.0157^{***}	0.0144^{***}	0.0050
	(0.00)	(0.00)	(0.00)	(0.000)	(0.182)	(0.000)	(0.000)	(0.331)
Book-to-market	-0.0286^{***}	-0.0289^{***}	-0.0288***	-0.0290^{***}	-0.0629^{***}	-0.0686^{***}	-0.0687***	-0.0521^{***}
	(0.00)	(0.00)	(0.00)	(0.000)	(0.00)	(0.00)	(0.00)	(0.00)
Bid-ask spread	-5.4485*	-4.6590^{***}	-4.6170^{***}	-4.7497^{***}	-4.0748	-5.9144^{**}	-5.6613^{**}	-7.8446^{**}
	(0.098)	(0.003)	(0.001)	(0.002)	(0.102)	(0.023)	(0.023)	(0.049)
Idio. volatility	-0.4787							5.8392^{***}
	(0.817)							(0.004)
Co-Skewness		0.0020						-0.0011
		(0.120)						(0.531)
Co-Kurtosis			0.0001					0.0001
			(0.205)					(0.166)
Downside beta				-0.0162				0.0245
				(0.742)				(0.573)
Volatility					-3.7612^{***}			-9.0684^{***}
					(0.001)			(0.000)
Skewness						0.0161^{***}		0.0286^{***}
						(0.007)		(0.000)
Kurtosis							0.0001	-0.0006***
							(0.691)	(0.002)
adj. R ²	0.1393	0.1246	0.1247	0.1237	0.1617	0.1464	0.1384	0.2018

	P.	anel B: Liquidity-Related Ch	aracteristics	
	(xviii)	(xix)	(xx)	(xxi)
Constant	-0.4056^{***}	-0.3998***	-0.4046^{***}	-0.4108***
	(0.000)	(0.000)	(0.000)	(0.000)
Beta	0.0280	0.0226	0.0200	0.0291
	(0.657)	(0.723)	(0.745)	(0.628)
dVIX	-0.2169^{**}	-0.2021^{**}	-0.1986^{**}	-0.2103^{**}
	(0.020)	(0.038)	(0.033)	(0.022)
dVVIX	-0.7673***	-0.6845^{**}	-0.6884^{**}	-0.7190^{**}
	(0.008)	(0.030)	(0.024)	(0.015)
ln(Size)	0.0340^{***}	0.0344^{***}	0.0319^{***}	0.0350^{***}
	(0.000)	(0.000)	(0.001)	(0.000)
Book-to-market	-0.0307^{***}	-0.0453^{***}	-0.0310^{***}	-0.0314^{***}
	(0.000)	(0.000)	(0.00)	(0.000)
PS liquidity	0.0121			
	(0.125)			
Amihud illiquidity		76.426***		
		(0.008)		
ln(Volume)			0.0021	
			(0.792)	
Turnover				-0.0317 (0.309)
adj. R ²	0.1310	0.1223	0.1225	0.1235

Table A3: Fama–MacBeth Regressions (Further Control Variables) (continued)

This table pres	ents average c	coefficient estim	ates from mon	thly value-weig	hted Fama &	MacBeth (1973) regressions.	Each month, we
regress excess ;	stock returns	during the follc	wing year on	a constant, the	e market betas	, the sensitivit	ies to innovatio	ons in aggregate
volatility, the se	ensitivities to i	nnovations in ag	ggregate volatil	ity-of-volatility	over the same	time, and/or a	series of $\beta_{j,t}^{\zeta}$ es	timates obtained
in a joint multi	variate regress.	ion as in Equat	ion (15) , using	daily data over	the following	year. Detailed \mathbf{v}	⁄ariable definiti	ons are provided
in Appendix A	of the main	paper. Robust	Newey & Wes	st (1987) p-val	ues using 12 la	gs, that also i	ncorporate the	Shanken (1992)
errors-in-variab	les correction,	are reported in	parentheses. *	, **, and *** ii	ıdicate significa	nce at 10%, 59	6, and 1%, resp	ectively.
	(i)	(ii)	(iii)	(iv)	(v)	(vi)	(vii)	(viii)
Constant	0.0638*	0.1031^{***}	0.0660^{**}	0.0933^{**}	0.0530	0.1002^{***}	0.0647^{**}	0.0513
	(0.057)	(0.003)	(0.040)	(0.013)	(0.117)	(0.005)	(0.048)	(0.112)
	0.562) (0.562)	-0.001)	0.0310	(0.982) (0.982)	0.0439 (0.440)	-0.0032 (0.931)	0.021) (0.576)	0.0407 (0.404)
dVIX	-0.2863^{***} (0.004)	-0.2490^{***} (0.008)	-0.2752^{***} (0.003)					-0.2740^{***} (0.002)
XIVVb	-0.7960***	-0.8274***	-0.8086***	-0.9663***	-0.8979***	-0.8593**	-0.8235***	-0.8793***
SMB	(0.009) -0.0076	(0.009)	(0.004) -0.0085	(0.006)	(0.004) - 0.0072	(0.010)	(0.006)-0.0075	(0.005)
	(0.791)		(0.754)		(0.796)		(0.790)	
TIMIT	(0.027)		(0.026)		(0.027)		(0.026)	
dSkew		0.7768*	0.6204*					
dKurt		-1.3639 -1.3639 -1.393	-1.0106					
Straddle vol		(071.0)	(191.0)	0.2748^{**}	0.1900^{***}			
Junp				(0.010) -1.4307***	(0.010)-1.1535***			
dVRP				(0.003)	(0.004)	0.0665^{**}	0.0748^{**}	
SMB*						(0.035)	(0.011)	-0.0561^{*}
V L								(0.079)
IA								-0.0064 (0.741)
ROE								0.0622^{***} (0.006)
adj. \mathbb{R}^2	0.2388	0.1937	0.2508	0.1691	0.2380	0.1737	0.2378	0.2556

Table A4: Fama–MacBeth Regressions – Multivariate Estimation (Value-Weighted)

Table A5: Realized Measure of Aggregate Volatility-of-Volatility

At the beginning of each month, we form equally weighted quintile portfolios based on the stock's sensitivities to innovations in aggregate volatility-of-volatility $(\beta_{j,t}^{Q})$, measured as the realized volatility of the VIX, over the following year. The extended sample period spans January 2000 until September 2016. To obtain the sensitivities, we regress daily excess stock returns on dVoVIX, controlling for MKT and dVIX as in Equation (14). Stocks with the lowest $\beta_{j,t}^{Q}$ are sorted into portfolio 1, those with the highest $\beta_{j,t}^{Q}$ into portfolio 5. The column labeled 5 minus 1 refers to the hedge portfolio buying the quintile of stocks with the highest $\beta_{j,t}^{Q}$ and simultaneously selling the stocks in the quintile with the lowest $\beta_{j,t}^{Q}$. Each month, we set up new 12-month portfolios. The row labeled Mean return is based on monthly simple returns. CAPM alpha, 4-factor alpha, and 5-factor alpha refer to the alphas of the CAPM, the Carhart (1997) 4-factor, and the 5-factor (including liquidity) models, respectively. HXZ alpha denotes the alpha relative to the Hou et al. (2015b) factor model. The segment Horizon 06-16 presents the results when restricting the sample period to the time frame between 2006 and 2016. Robust Newey & West (1987) p-values using 12 lags are reported in parentheses. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

Rank	1	2	3	4	5	5 minus 1
Mean return	0.0798	0.1004***	0.1019^{***}	0.0961^{**}	0.0457	-0.0341
	(0.100)	(0.007)	(0.000)	(0.025)	(0.402)	(0.180)
CAPM alpha	0.0161	0.0547^{***}	0.0578^{***}	0.0453^{**}	-0.0280	-0.0441*
	(0.224)	(0.004)	(0.005)	(0.032)	(0.259)	(0.080)
4-factor alpha	-0.0008	0.0251^{***}	0.0235***	0.0100^{*}	-0.0647***	-0.0639***
	(0.956)	(0.000)	(0.000)	(0.086)	(0.000)	(0.003)
5-factor alpha	-0.0052	0.0188^{***}	0.0200***	0.0102^{**}	-0.0499***	-0.0447***
	(0.663)	(0.000)	(0.000)	(0.047)	(0.000)	(0.006)
HXZ alpha	-0.0088	0.0095	0.0089^{*}	0.0088	-0.0366**	-0.0278
	(0.330)	(0.128)	(0.072)	(0.183)	(0.039)	(0.234)
Horizon 06-16						
Mean return	0.0978	0.0978^{*}	0.0964*	0.0926	0.0548	-0.0430
	(0.111)	(0.062)	(0.071)	(0.129)	(0.489)	(0.249)
HXZ alpha	0.0051	0.0204***	0.0149**	0.0068	-0.0559***	-0.0610***
	(0.553)	(0.000)	(0.021)	(0.396)	(0.000)	(0.004)